### **Oracle® Siebel Retail Finance**

Branch Teller Guide
Release 8.1.1 for Siebel Branch Teller **E20122-01** 

March 2011



Oracle® Siebel Retail Finance Branch Teller Guide, Release 8.1.1 March 2011

F20122-01

Copyright © 2005, 2011, Oracle and/or its affiliates. All rights reserved.

The Programs (which include both the software and documentation) contain proprietary information; they are provided under a license agreement containing restrictions on use and disclosure and are also protected by copyright, patent, and other intellectual and industrial property laws. Reverse engineering, disassembly, or decompilation of the Programs, except to the extent required to obtain interoperability with other independently created software or as specified by law, is prohibited.

The information contained in this document is subject to change without notice. If you find any problems in the documentation, please report them to us in writing. This document is not warranted to be error-free. Except as may be expressly permitted in your license agreement for these Programs, no part of these Programs may be reproduced or transmitted in any form or by any means, electronic or mechanical, for any purpose.

If the Programs are delivered to the United States Government or anyone licensing or using the Programs on behalf of the United States Government, the following notice is applicable:

#### U.S. GOVERNMENT RIGHTS

Programs, software, databases, and related documentation and technical data delivered to U.S. Government customers are "commercial computer software" or "commercial technical data" pursuant to the applicable Federal Acquisition Regulation and agency-specific supplemental regulations. As such, use, duplication, disclosure, modification, and adaptation of the Programs, including documentation and technical data, shall be subject to the licensing restrictions set forth in the applicable Oracle license agreement, and, to the extent applicable, the additional rights set forth in FAR 52.227-19, Commercial Computer Software--Restricted Rights (June 1987). Oracle USA, Inc., 500 Oracle Parkway, Redwood City, CA 94065.

The Programs are not intended for use in any nuclear, aviation, mass transit, medical, or other inherently dangerous applications. It shall be the licensee's responsibility to take all appropriate fail-safe, backup, redundancy and other measures to ensure the safe use of such applications if the Programs are used for such purposes, and we disclaim liability for any damages caused by such use of the Programs.

The Programs may provide links to Web sites and access to content, products, and services from third parties. Oracle is not responsible for the availability of, or any content provided on, third-party Web sites. You bear all risks associated with the use of such content. If you choose to purchase any products or services from a third party, the relationship is directly between you and the third party. Oracle is not responsible for: (a) the quality of third-party products or services; or (b) fulfilling any of the terms of the agreement with the third party, including delivery of products or services and warranty obligations related to purchased products or services. Oracle is not responsible for any loss or damage of any sort that you may incur from dealing with any third party.

Oracle, JD Edwards, PeopleSoft, and Siebel are registered trademarks of Oracle Corporation and/or its affiliates. Other names may be trademarks of their respective owners. This documentation is in prerelease status and is intended for demonstration and preliminary use only. It may not be specific to the hardware on which you are using the software. Oracle Corporation and its affiliates are not responsible for and expressly disclaim all warranties of any kind with respect to this documentation and will not be responsible for any loss, costs, or damages incurred due to the use of this documentation. The information contained in this document is for informational sharing purposes only and should be considered in your capacity as a customer advisory board member or pursuant to your beta trial agreement only. It is not a commitment to deliver any material, code, or functionality, and should not be relied upon in making purchasing decisions. The development, release, and timing of any features or functionality described in this document remains at the sole discretion of Oracle.

This document in any form, software or printed matter, contains proprietary information that is the exclusive property of Oracle. Your access to and use of this confidential material is subject to the terms and conditions of your Oracle Software License and Service Agreement, which has been executed and with which you agree to comply. This document and information contained herein may not be disclosed, copied, reproduced, or distributed to anyone outside Oracle without prior written consent of Oracle. This document is not part of your license agreement nor can it be incorporated into any contractual agreement with Oracle or its subsidiaries or affiliates.

# Contents

Pr	eface		. vi
	Audie	nce	vi
	Docun	nentation Accessibility	vi
		d Documents	
	Conve	ntions	vii
	Conte	nts in this Guide	vii
1	What's	s New In This Release	
	1.1	What's New in Siebel Branch Teller, Version 8.1.1	1-1
2	Overv	iew	
	2.1	The Siebel Branch Teller Application	2-1
	2.2	Navigating Through the Siebel Branch Teller Application	
	2.2.1	Navigating Tables	
	2.3	The Operations Home Page	
	2.4	Customer Sessions and Incomplete Customer Sessions	
	2.5	Main Menu, Transaction Area, and Status Bar	2-4
	2.6	The Customer Relationship Console	2-4
	2.7	Date Formats	2-5
	2.8	Peripheral Devices	2-5
3	Gettin	g Started with Siebel Branch Teller	
	3.1	Starting the Application	3-1
	3.2	Logging In to Siebel Branch Teller	3-1
	3.2.1	User Session Timeout	3-2
	3.3	Standard Buttons and Keys in Siebel Branch Teller	3-2
	3.3.1	Short Cut keys	3-3
	3.3.2	Keyboard Shortcuts	3-3
	3.4	Online and Offline Processing Modes	3-4
	3.5	AM and AM/PM Processing Modes	3-5
	3.6	Account-Centric Mode and Customer-Centric Processing Modes	3-5
	3.7	Supervisor Override	3-5
	3.8	Performing a Local Supervisor Override	3-6
	3.9	Performing a Remote Supervisor Override	3-7
	3.10	Temporary Logoff	3-7

	3.11	Logging Off	3-7
4	Searc	hing Customer Details and other commonly used features	
	4.1	Searching Customer Information	4-1
	4.2	Incomplete Customer Sessions	
	4.3	Logging Out Temporarily	4-2
	4.4	The Customer Relationship Console	4-2
	4.4.1	Viewing Customer Summary Information	4-3
	4.4.2	Viewing Customer Account Information	4-3
	4.4.3	Viewing Customer Offer Information	4-4
	4.4.4	Creating a Referral	
	4.4.5	Viewing Customer Alert Information	4-5
	4.4.6	Viewing Service Request Information	4-5
5	Perfo	rming Customer Financial Transactions	
	5.1	Customer Sessions	5-2
	5.1.1	Reversing Transactions	5-3
	5.2	Currency Transaction Reporting	5-4
	5.2.1	Completing the Currency Transaction Report	5-4
	5.2.2	Resuming a Suspended Currency Transactions Report	5-5
	5.2.3	Printing Currency Transaction Report Form	5-6
	5.3	Common Tasks for Financial Transaction Operations	5-6
	5.3.1	Selecting the Account	5-6
	5.3.2	Entering Check Details	5-7
	5.3.3	Entering Payment Method Details	5-7
	5.3.4	Waiving the Fee for a Transaction	5-8
	5.3.5	Entering Customer Authentication Details	5-8
	5.3.6	Receipt Printing Transactions	5-8
	5.4	Making a Deposit	5-9
	5.5	Making a Withdrawal	. 5-10
	5.6	Cashing a Check	
	5.7	Transferring Funds Between a Customer's Accounts	. 5-11
	5.8	Transferring Funds to Another Account	
	5.9	Making a Wire Transfer	. 5-12
	5.10	Making International Transfer	. 5-13
	5.11	Making a Payment to a Utility	. 5-14
	5.12	Making a Cash Denomination Exchange	. 5-15
	5.13	Making a Foreign Currency Exchange	. 5-15
	5.14	Issuing a Local Draft	. 5-15
	5.15	Issuing a Draft in a Foreign Currency	. 5-16
	5.16	Issuing a Cashier's Check	. 5-17
	5.17	Issuing a Money Order	. 5-18
	5.18	Purchasing U.S. Savings Bonds	. 5-18
	5.19	Making U.S. Savings Bond Inquiries and Redeeming Bonds	

6	Perfo	rming Customer Inquiries	
	6.1	Displaying Account History	6-1
	6.2	Maintaining Transfers	6-2
	6.3	Maintaining Payments	6-2
	6.4	Displaying Check Details and Stopping Checks	6-3
	6.5	Displaying Credit Card Movements	6-4
	6.6	Displaying the Currency Exchange Rates	6-4
	6.7	Displaying the Interest Rates	6-4
7	Perfo	rming Customer Services	
	7.1	Maintaining Customer Details	
	7.2	Maintaining Beneficiary Details	7-2
	7.3	Maintaining Utility Payment Details	
	7.4	Maintaining Standing Orders	
	7.4.1	To maintain standing orders	7-4
	7.5	Maintaining Direct Debits	7-5
	7.6	Maintaining Overdrafts	7-6
	7.7	Maintaining Statements	7-7
	7.8	Maintaining Bank Cards	7-7
	7.9	Issuing a Checkbook	
	7.10	Opening an Account	7-9
	7.11	Closing an Account	7-11
	7.12	Creating a Credit Card Application	7-12
8	Perfo	rming Teller Operations	
	8.1	Starting Your Day	8-1
	8.1.1	To start your day	8-2
	8.1.2	Available Teller Transactions	8-4
	8.2	Performing Interim Balance Inquiries	8-5
	8.3	Performing Batch Inquiries	8-5
	8.4	Opening Cashboxes/Drawers during Signon	8-6
	8.5	The Journal Activities	8-6
	8.5.1	Displaying the Journal, Reversing Transactions, and Printing a session receipt	8-8
	8.6	Reconciling Failed Transactions	8-10
	8.7	Debiting or Crediting a General Ledger Account	8-10
	8.8	Transferring Cash Between Cashboxes	8-11
	8.9	Transferring Checks Between Cashboxes	8-12
	8.10	Transferring Ownership of a Cashbox	8-12
	8.11	Closing Open Batches of Checks	8-13
	8.12	Transferring a Checkbook	8-14
	8.13	Ending the Business Day	8-14
	8.14	Performing End-of-Day Processing	8-15
	8.15	Logging Out	8-16
9	Perfo	rming Branch Operations	
	9.1	Starting the Branch Day	9-1

9.2	Searching Branch Information	9-2
9.3	Performing Cashbox Maintenance	9-2
9.4	Performing Vault Maintenance	9-3
9.5	Making External Cash Transfers	9-4
9.6	Making External Check Transfers	9-4
9.7	Resetting the Login Counter	9-5
9.8	Forcing Sign Off	9-6
9.9	Ending the Branch Business Day	9-6
9.10	Ending the Branch Day	9-6

# **Preface**

Siebel Branch Teller is a retail banking application that gives a 360 degree view of the customer and financial transactions of the customer to the customer-facing personnel (teller) of the bank. This enables you as the teller, to provide better customer focused services as well as cross sell and up sell the other products and services of the bank. This user guide is directed to help you with required information that enables you to perform well with Siebel Branch Teller.

### **Audience**

This document is intended for the bank tellers and teller supervisors.

## **Documentation Accessibility**

Our goal is to make Oracle products, services, and supporting documentation accessible to all users, including users that are disabled. To that end, our documentation includes features that make information available to users of assistive technology. This documentation is available in HTML format, and contains markup to facilitate access by the disabled community. Accessibility standards will continue to evolve over time, and Oracle is actively engaged with other market-leading technology vendors to address technical obstacles so that our documentation can be accessible to all of our customers. For more information, visit the Oracle Accessibility Program Web site at http://www.oracle.com/accessibility/.

#### Accessibility of Code Examples in Documentation

Screen readers may not always correctly read the code examples in this document. The conventions for writing code require that closing braces should appear on an otherwise empty line; however, some screen readers may not always read a line of text that consists solely of a bracket or brace.

#### Accessibility of Links to External Web Sites in Documentation

This documentation may contain links to Web sites of other companies or organizations that Oracle does not own or control. Oracle neither evaluates nor makes any representations regarding the accessibility of these Web sites.

#### **Access to Oracle Support**

Oracle customers have access to electronic support through My Oracle Support. For information, visit http://www.oracle.com/support/contact.html or visit http://www.oracle.com/accessibility/support.html if you are hearing impaired.

### **Related Documents**

Oracle's Siebel Branch Teller Bookshelf is available on Oracle Technology Network at

http://www.oracle.com/technetwork/index.html. For more information, see the following documents on Siebel Bookshelf on Oracle Technology Network (OTN):

■ Siebel Retail Finance Online Documentation Library Release 8.1.1

### **Conventions**

The following text conventions are used in this document:

Convention	Meaning
boldface	Boldface type indicates graphical user interface elements associated with an action, or terms defined in text or the glossary.
italic	Italic type indicates book titles, emphasis, or placeholder variables for which you supply particular values.
monospace	Monospace type indicates commands within a paragraph, URLs, code in examples, text that appears on the screen, or text that you enter.

# **Contents in this Guide**

Chapter Name	Description
Chapter 1, "What's New In This Release"	This chapter gives information about the changes that occurred in the 8.1.1 version of Siebel Branch Teller.
Chapter 2, "Overview"	This chapter provides an overview of Oracle's Siebel Branch Teller application.
Chapter 3, "Getting Started with Siebel Branch Teller"	This chapter provides an introduction to the Siebel Branch Teller application, and explains key terms and concepts that are used in the application and in this guide.
Chapter 4, "Searching Customer Details and other commonly used features"	This chapter describes the commonly used features that are listed in the top left side menu that appear as static menu, always visible to you. It also describes details about the Customer relationship Console.
Chapter 5, "Performing Customer Financial Transactions"	This chapter describes the financial transactions that you perform for the customers. When you select one of the operations, a customer session starts, and you can then perform a number of transactions in the session.
Chapter 6, "Performing Customer Inquiries"	This chapter describes the customer inquiries that you perform to obtain more information about the customer accounts.
Chapter 7, "Performing Customer Services"	This chapter describes the customer services operations that you perform to maintain detail of customers, beneficiaries, and utility payments applying to their accounts.
Chapter 8, "Performing Teller Operations"	This chapter describes the teller operations that you perform that do not involve customer accounts.
Chapter 9, "Performing Branch Operations"	This chapter describes the branch operations that you perform to start and end the branch day, set up cashboxes, and control the logging in of users.

Chapter Name	Description
Chapter 10, "Installing Oracle BI Publisher Desktop"	This chapter describes the activities you need to perform to install Business Intelligence Publisher (BI Publisher) Desktop

# What's New In This Release

This chapter gives information about the changes that occurred in the 8.1.1 version of Siebel Branch Teller.

# 1.1 What's New in Siebel Branch Teller, Version 8.1.1

The following table lists changes in this version of the documentation to support version 8.1.1 of the software.

What's New in Siebel Branch Teller, Version 8.1.1

Topic	Description
Menu Reorganization	The Teller menu structure has been reorganized in this release. Correspondingly the user guide is reorganized to reflect the changes in the menu during this release of Siebel Branch Teller.
Java Documentation	Java documentation has been introduced in this release to replace Rational Rose and Siebel Design tools. The earlier HTML design documents created and maintained using Rational Rose have been replaced with HTML Documents generated through running "JAVADOC" (via a supplied build.xml) on generated Java files.
Business Process Models	Business Process Models of the Teller functionality have been added in this release.
Capture Contact Event	This new feature enables the Siebel Teller application to send a notification to Siebel CRM Financial Services 8.1.1 when there has been a customer interaction in Siebel Branch Teller. This information is sent asynchronously from Siebel Branch Teller to Siebel CRM as an Web Service request when a customer session is complete, cancelled or suspended. This information will include summary details of the transaction performed (for example, Description, Type, Duration, Site, and so on of the session) and these details will be stored under the <b>Activities</b> tab in Siebel.
Currency Transaction Reporting	Currency Transaction Reporting (CTR) has been updated to leverage Oracle BI Publisher. This new feature enables Oracle BI Publisher 10.1.3 to be used for the design and generation of CTR reports in PDF, in replacement of Accellio. CTR details are captured and sent to BI Publisher via a Web Service call, which in turn generates a report that is stored. Tellers and Branch administrators can search and browse for completed or suspended CTR reports, and then open these reports in their Internet browser, as a PDF, for review and printing.

Торіс	Description
Translation Support	Spanish, Portuguese, and Russian language support has been added in this release
Privileges and the Entitlements Operations	To perform many of the Siebel Branch Teller operations, you must have the necessary privileges. For example, to perform the Create Cashbox operation, you must have supervisor privileges.
	An administrator uses the Entitlements operations to define the privileges. The administrator can define privileges to perform particular tasks (operations), and can also define privileges in terms of roles. The Entitlements operations therefore allows the administrator to define the roles and associated privileges that match your organization. That is, roles such as teller, supervisor, manager, administrator, and so on.
	The Entitlements operations also allow the setting of limits, which are used, for example, to control the value of transactions that tellers can perform.
	For more information about the Entitlements operations, see the <i>Siebel Retail Finance Banking Application Administration Guide</i> .
Encryption	The password and CTR report ids are encrypted in Siebel Branch Teller. That is, the CTR report id won't appear in the URL. Therefore, this document is URL protected.
Section 3.2.1, "User Session Timeout"	Updated the information related to User Session Timeout due to inactivity.
Section 3.7, "Supervisor Override"	Updated fields for multiple approval reasons for supervisor override. Modifications in both Local and Remote Supervisor Overrides.
Section 4.2, "Incomplete Customer Sessions"	The incomplete customer sessions are displayed and can be completed.
Section 5.9, "Making a Wire Transfer"	Updated the wire transfer details in Customer Transactions > Wire Transfer.
Section 5.10, "Making International Transfer"	Updated the wire transfer details in Customer Transactions > International Transfer.
Section 6.7, "Displaying the Interest Rates" and Section 6.6, "Displaying the Currency Exchange Rates"	The interest rates and currency exchange rates are available for viewing in Siebel Branch Teller.
Section 7.4.1, "To maintain standing orders"	Updated the information regarding specification of an end date for suspending a standing order instruction.
Table 8.1, "Starting Your Day"	Updated the information related to cashless teller who assumes the charge to help a teller who handles cash transactions.
Table 8.1, "Starting Your Day"	Updated the information related to cashless teller who can perform cash related transactions. The incomplete customer session is transferred to a teller with a cashbox to complete the session.
Section 8.5, "The Journal Activities"	Updated the information related to the changes in Journal Summary Fields.
Section 8.5.1, "Displaying the Journal, Reversing Transactions, and Printing a session receipt"	Updated the information related to printing a session receipt. The session receipt can be reprinted for the customer or in case of any printing error during automatic printing.

Торіс	Description
Section 8.7, "Debiting or Crediting a General Ledger Account"	Updated the information related to the changes in the General Ledger transactions.

What's New in Siebel Branch Teller, Version 8.1	ler. Version 8.1	Teller.	Branch	Siebel	New in	What's
---	------------------	---------	--------	--------	--------	--------

# **Overview**

This chapter provides an overview of Oracle's Siebel Branch Teller application. It includes the following topics:

- Section 2.1, "The Siebel Branch Teller Application"
- Section 2.2, "Navigating Through the Siebel Branch Teller Application"
- Section 2.3, "The Operations Home Page"
- Section 2.4, "Customer Sessions and Incomplete Customer Sessions"
- Section 2.5, "Main Menu, Transaction Area, and Status Bar"
- Section 2.6, "The Customer Relationship Console"
- Section 2.7, "Date Formats"
- Section 2.8, "Peripheral Devices"

### 2.1 The Siebel Branch Teller Application

The Siebel Branch Teller operations are grouped into the following menus:

- Customer Search, Incomplete Sessions, and Temporary Logoff. Operations that are always visible to you for easier access appears first.
- Customer Transactions. Operations for performing the financial transactions for customer accounts, for example, deposits and withdrawals.
- Customer Services. Operations for opening, maintaining, and closing customer accounts and ancillary services.
- Customer Inquiries. Operations pertaining to customer account transactions and bank rates information.
- Teller Operations. Operations that you need to start your day, maintain your cash and currency boxes, and end your day.
- Branch Operations. Operations that you need to end branch processing for the day, set up cashboxes, and control the logging in of users.

In each application menu the operations are numbered. You can select an operation either by clicking the menu and then the operation, or by typing a number in the Fast Tran field. For example, to select the Deposit operation, type 21 in the Fast Tran field.

You have the ability to change the default sort order of a list by selecting the column header in a list.

The tasks that you can perform with the operations are described in the rest of the chapters.

### 2.2 Navigating Through the Siebel Branch Teller Application

You can navigate through the Siebel Branch Teller application either by using the tab key, or by clicking with the mouse.

Generally, when you navigate using tab key, you go from top to bottom and left to right through a screen.

You can select radio buttons either by using the mouse or by using the spacebar.

### 2.2.1 Navigating Tables

In Siebel Branch Teller, you can navigate through tables as follows:

- Use the up and down arrow keys to move up and down the rows in the table.
- Use the tab key to move between the cells in the table, from the top left to the bottom right.
- Use the left and right arrow keys to move left and right within the rows of the
- Click on the first cell of a row to select a row.
- Press CTRL+TAB to exit from a table, when you have not selected a row, or are not at the last row that contains data.

### 2.3 The Operations Home Page

When you are not performing an operation, the operations home page is displayed. On the left side of the screen are the menus of operations that you can perform. In the Fast Tran field you can type a two or three-digit number and press Enter to quickly go to the operation corresponding to that number.

Following table describes the fields displayed under Transaction Details.

Operations Home Page Fields

Field	Comment
User ID	Displays your user ID.
Today's Date	Displays the calendar date.
Processing Date	Displays the processing date. If your branch is operating in AM/PM processing mode, this date is different from the calendar date in the P.M. period. See Section 3.5, "AM and AM/PM Processing Modes", for more information.

**Note:** If there are incomplete sessions from your previous login, that information will appear under Transaction Details. It will be the following message in red: You currently have incomplete sessions. Please select "Incomplete Sessions" from the inquiries menu to complete these sessions.

The following table lists the various indicators in the status line at the bottom right of the screen.

Indicators on Operations Home Page

Indicator		Comment
	<b>     </b>	Indicates whether customer details are retrieved during financial transactions. Click the indicator to switch this facility on and off. See Section 3.6, "Account-Centric Mode and Customer-Centric Processing Modes", for more information.
	BUSINESS DAY: 02/01/2005	Indicates the current processing date.
	SERMER: Online	Indicates whether Siebel Branch Teller is online. That is, whether your Siebel Branch Teller application is communicating with the central server. See Section 3.4, "Online and Offline Processing Modes", for more information.
	CTR	Indicates that there are outstanding currency transaction reports (CTR) to be processed. When you hover over this icon with the mouse, the following message is displayed:
		You have suspended CTRs. Selecting this icon will launch these CTRs from where you may resume processing.
		On Clicking CTR icon, the journal search screen will be displayed and teller needs to perform journal search by choosing CTR Suspended from the Transaction Type drop down list. The journal summary will retrieve the suspended CTRs.
		For more information about currency transaction reporting, see Section 5.2, "Currency Transaction Reporting".
	<b>A</b>	Indicates that there are outstanding override requests for the supervisor that is to be addressed. This flashing indicator comes when the teller has submitted a remote supervisor override request. For more information about remote supervisor request, see Section 3.9, "Performing a Remote Supervisor Override".

As you perform operations, Siebel Branch Teller displays messages if you perform an invalid action.

### 2.4 Customer Sessions and Incomplete Customer Sessions

A customer session starts, when you begin a financial transaction. You can perform various transactions within a customer session. For more information on Customer Session, see Section 5.1, "Customer Sessions".

A customer session becomes an incomplete customer session, when it is paused unintentionally or intentionally. The incomplete sessions in Active or Suspended status are displayed at Incomplete Sessions (Fast Tran 12). For more information on Incomplete Customer Session, see Section 4.2, "Incomplete Customer Sessions".

### 2.5 Main Menu, Transaction Area, and Status Bar

The general layout for all Siebel Branch Teller consists of Customer Relationship Console, Main Menu, Transaction Area, and Status Bar. For more information on Customer Relationship Console, see, Section 2.6, "The Customer Relationship Console" and Section 4.4, "The Customer Relationship Console".

The First Level Main Menu groups the main areas, for example, Customer Inquiries, Customer Transactions, and so on. Within each Main Menu, the areas are further categorized into specific functions that are represented by menu items in the left navigation bar.

The Transaction Area consists of Transaction Panel and Transaction Summary Panel.

The Status Bar is present in the bottom. It displays some helpful information that you can control. For example, you can click the customer-centric/account-centric indicator on the status bar at the bottom of the screen depending on whether you want to receive customer details.

### 2.6 The Customer Relationship Console

Your Siebel Branch Teller application might be integrated with the Siebel Financial Services 8.0 or 8.1.1 application so that customer relationship management information is available to you. This means that:

- When you search for customer information, you get a complete profile of the customer.
- When you perform customer transactions, you can provide offers, alerts, referrals, and other value-added services to the customer. This allows you to provide a better service to the customer and sales opportunities for your bank.

The information appears in a customer relationship console (CRC) that Siebel Branch Teller displays when you complete a customer search operation, and when you perform account maintenance and customer maintenance operations in what is called customer-centric mode.

The CRC contains the following areas:

- Summary. Contains the customer's personal information, financial profile information, and a summary of offers and alerts related to the customer
- **Account.** Contains details of the customer's accounts and the status of the accounts
- **Offers.** Contains details of the offers that are currently available to the customer, and allows you to make referrals based on those offers
- **Alerts.** Contains details of information about which you should inform the customer

- **SRs.** Contains service requests relevant to the customer, for your information only
- Signature. Contains the scanned signature of the customer that can be obtained clicking this button, seen below the photograph of the customer.

For more information about the CRC, see Section 4.4, "The Customer Relationship Console".

#### 2.7 Date Formats

The format in which you enter dates is determined when the Siebel Branch Teller application is installed. The date format that your bank uses is determined by your location. In the U.S, the date format is likely to be MM/DD/YYYY, for example, and you would enter 06/12/2005 for the 12th of June 2005. In Europe, the date format is likely to be DD/MM/YYYY, for example, and you would enter 12/06/2005 for the 12th of June 2005.

For some operations you must enter the date in a specific format. The format is described in the sections for those operations in this book.

### 2.8 Peripheral Devices

Siebel Branch Teller supports the following peripheral devices:

- MagTek MagStripe and PIN Pad
- Epson receipt printer
- MagTek Magnetic Ink Character Recognition (MICR) check reader
- Source Technologies MICR printer

You use the MagStripe and PIN pad device to read the data on the magnetic strip on the back of a card. The data retrieved depends on what type of card you swipe, but typically includes branch code, card number, and account name. When you swipe the card, the customer is prompted to type a Personal Identification Number (PIN) on the keypad provided.

# **Getting Started with Siebel Branch Teller**

This chapter provides an introduction to the Siebel Branch Teller application, and explains key terms and concepts that are used in the application and in this guide. It includes the following topics:

- Section 3.1, "Starting the Application"
- Section 3.2, "Logging In to Siebel Branch Teller"
- Section 3.3, "Standard Buttons and Keys in Siebel Branch Teller"
- Section 3.4, "Online and Offline Processing Modes"
- Section 3.5, "AM and AM/PM Processing Modes"
- Section 3.6, "Account-Centric Mode and Customer-Centric Processing Modes"
- Section 3.7, "Supervisor Override"
- Section 3.8, "Performing a Local Supervisor Override"
- Section 3.9, "Performing a Remote Supervisor Override"

### 3.1 Starting the Application

The Siebel Branch Teller application and other software required by Siebel Branch Teller will have been installed on your workstation. To start the Siebel Branch Teller application, click the icon for the application on your desktop.

## 3.2 Logging In to Siebel Branch Teller

You log in to Siebel Branch Teller by entering your user ID, password, and branch number. Note that the user ID and password are case sensitive. Siebel Branch Teller then checks that your user ID and password are valid. If you try to log in unsuccessfully a number of times, you are locked out of the system. You will remain locked out until the supervisor resets the login counter for you.

If you are a supervisor and it is the start of the branch day, you must confirm the branch start of day totals. For more information, see Section 9.1, "Starting the Branch Day".

If you are a teller logging in for the first time in the business day, you must perform a Teller Start of Day operation and be assigned a cashbox, or you can carry on with your previously assigned cashbox. You must reconcile your cashbox at this time. You can also login as a Cashless User (Disable Cashbox Transactions) or a Cashless User (Enable Cash Transactions). You can then perform operations depending on the privileges

assigned to your role. The operations available to you on the application menu are determined by your role.

Before you can log out at the end of the day, you must perform a Teller End of Day operation.

#### 3.2.1 User Session Timeout

This feature enables you to receive a pop up warning that your customer session is about to expire. Depending on your response, the customer session can be continued or allowed to time out.

If you are inactive for a defined period, Siebel Branch Teller informs you that you will be logged out within a specific time interval. You can return to Siebel Branch Teller by clicking the Cancel button. If you did not click the Cancel button within that time interval, you will be temporarily logged out and will be directed to the logon screen. When you need to take a break, you can log out temporarily, and your workstation is then available for another teller to use.

### 3.3 Standard Buttons and Keys in Siebel Branch Teller

Details of what the overall style of application usage is, within the Siebel Branch Teller, can be found in the Siebel Branch Teller Style Guide. Refer to this document too.

In Siebel Branch Teller, buttons have consistent actions as described in the following table.

Standard Buttons in Siebel Branch Teller

Button	Action	
Amend	A screen allowing you to change details is displayed.	
Approve	Sanction a request.	
Cancel	All the input values are cleared and you are returned to the Home page.	
Cancel Session	All processing that has taken place is reversed and you are returned to the Home page.	
Clear	Your input is cleared. The same screen is then displayed, with fields in their default state.	
Close	Close the dialog box with an option to save data.	
Decline	Deny a request.	
End	The customer session ends, indicating that there are no further transactions in this customer session. A combined receipt is printed for all transactions executed during the customer session.	
New	New details are created.	
Next	For navigation to next page in a list.	
Previous	For navigation to previous page in a list.	
Print	A receipt or other information is printed.	
OK	You exit from an information page.	
Reverse	The selected transaction is reversed.	
Search	A search is performed using the search criteria that you specified.	
New Next Previous Print OK Reverse	The customer session ends, indicating that there are no further transactions in this customer session. A combined receipt is printed for all transactions executed during the customer session.  New details are created.  For navigation to next page in a list.  For navigation to previous page in a list.  A receipt or other information is printed.  You exit from an information page.  The selected transaction is reversed.	

Button	Action
Submit	The transaction is saved and processed.
Suspend	The status of Customer Session is set as Suspended and session can be resumed later to complete the transaction/operation. After ending the Customer Session, you are directed to the Home page.
VR	The virtual cash amount is added to the transaction amount field.

You can select the buttons by using accelerator keys. In each screen, the names of the buttons will have an underscore beneath one of their letters. Pressing the ALT key together with the key is the same as selecting the button with the mouse. For example, on most screens, you can perform a Cancel function, by pressing ALT+C together.

### 3.3.1 Short Cut keys

In Siebel Branch Teller, various activities performed using a mouse can be done using short cut keys. The short cut keys are numbered according to the number in which the menu items and sub menu items appear. For example, Customer Transactions is the second menu item and Deposit is the first sub menu item in that. Therefore 21 is the short cut key for the same. Typing 21 and pressing Enter in the Fast Tran text field takes you to Customer Transactions > Deposit.

### 3.3.2 Keyboard Shortcuts

In Siebel Branch Teller, most of the function keys (F1 through F12) have the same effect as clicking a particular button, as shown in the following table.

Keyboard Shortcuts in Siebel Branch Teller

Keyboard Shortcuts	Equivalent Button	Comment
F1		No keyboard shortcut. Reserved for online help.
F2	Search	A search is performed using the search criteria that you specified.
F3	Clear	Your input is cleared. The same screen is then displayed, with fields in their default state.
F4	Customer Search	Searches the details about the customer.
F5		No keyboard shortcut.
F6	Close	You close, when you are finished with the activity.
F7	Cancel	The cursor is returned to the Fast Tran field.
F8	Submit	The transaction is saved and processed.
F9	Amend	A screen allowing you to change details is displayed.
F10	New	New details are created.
F11		No keyboard shortcut.
F12	Delete	You delete a beneficiary or a pending payment, or so on according to your requirement.

### 3.4 Online and Offline Processing Modes

In online processing, your teller workstation communicates with a central server computer, which in turn communicates with a host system that performs processing and contains databases relevant to some transactions. In online mode, transactions are processed as normal, but when communication with the central server is lost, the branch enters offline processing mode.

The Siebel Branch Teller provides an indicator at the bottom right of the screen, indicating whether the Branch is currently working in Online or Offline processing mode. This indicator changes to Offline as soon as the branch loses communication with the central server.

The offline server keeps pinging or contacting the central server in periodic intervals. After the contact has been established, the offline server will forward the offline requests to the central server. The failed transactions are available for reconciliation.

A sub-set of Siebel Branch Teller operations are available in offline processing mode. For the operations that you can perform in offline processing mode, transactions are stored and later forwarded to the central server, and on to the host system, when online processing is reestablished. The host system can accept, or reject transactions for a variety of reasons, for example, because you performed the transactions against a closed account, or because there are insufficient funds.

Different business rules can apply for offline processing mode, for example, transaction limits can vary between online and offline processing mode.

At the end of the processing day, you have to reconcile transactions that have been rejected during forwarding. You use the Reconciling operation for reconciling such transactions.

The following Branch operations are available in offline processing mode:

- Branch Start Of Day
- Create Cashbox
- External Cash Transfer
- **External Check Transfer**
- Vault Maintenance
- Branch End of day
- Branch End of Business day

The following Teller operations are available in offline processing mode:

- Logon
- Temporary Logoff
- Teller Start Of Day
- Check Batch
- **Batch History**
- Cashbox Takeover
- Cashbox Check Transfer
- **Journal**
- Interim Balance

- Cashbox Cash Transfer
- Teller End Of Day
- Teller End Of business Day
- Logoff

The following financial transactions are available in offline processing mode:

- Deposit
- Withdrawal
- Cash Exchange

### 3.5 AM and AM/PM Processing Modes

It is a common practice for financial institutions to switch to the next business day during a calendar day. The branch processing mode defines whether or not the branch switches the transaction processing date to the next business day during a calendar day. There are two processing modes:

- In AM processing mode, Siebel Branch Teller does not switch the processing date to the next business day during a calendar day. All transactions are performed and posted on the same date. In AM processing mode, the processing date is the same as the calendar date.
- In AM/PM processing mode, transactions performed in the A.M. period are posted on the calendar date. At the start of the P.M. period, you perform a Branch End of Day operation to set the processing date to the next business day. Tellers can then perform transactions in the P.M. period and the transactions are posted on the next business day.

You set the AM and AM/PM processing mode for a branch in the Administration Console. This is done when the branch is closed. For more information about using the Administration Console, see the Siebel Retail Finance Banking Application Administration Guide.

### 3.6 Account-Centric Mode and Customer-Centric Processing Modes

You can choose whether to retrieve customer details when you are performing financial transactions. By default, customer details are retrieved, and the customer-centric mode indicator displays a green tick sign.

If you click the indicator, it changes to a red stop-sign, and customer details are not retrieved during the transaction; this is referred to as account-centric mode (or fast transaction processing mode). At the end of the transaction, Siebel Branch Teller reverts to customer-centric mode.

# 3.7 Supervisor Override

Supervisor Override allows Siebel Branch Teller to enforce a second user validation. There can be different reasons for a supervisor override like an operation above the user's monetary limit or a business rule that has been triggered to require a supervisor override.

When you perform an operation, Siebel Branch Teller checks your privileges and, where applicable, your limits for performing the operation. A single transaction may require an override for multiple reasons. For example, you might be able to make

withdrawals only up to \$1000 and you cannot request a cash back amount for a check that is not "On us". There are limits both for particular products and for you as a user.

In cases where you cannot proceed with an operation, Siebel Branch Teller advises you of the reason and that you require a supervisor override to continue. You can then request a supervisor override locally or remotely in order to proceed with the transaction. Siebel Branch Teller notifies your supervisor of the privileges and limits that you have exceeded together with the relevant transaction information.

Supervisors can approve or decline the override request. A supervisor can do this either at your workstation (local supervisor override) or from his or her own workstation, possibly in a different branch in the same region (remote supervisor override), in which case a response is returned to you. If a supervisor does not reply within a set time, the teller has to re-send the override request or cancel it. While Awaiting Supervisor Assignment or Awaiting Supervisor Response, a countdown timer is presented to you with an option to cancel.

When a single transaction requires override for multiple reasons, the supervisor should approve all the reasons for approving the override request.

Supervisor Override requests are routed to supervisors in various branches across the same region. The supervisor is able to view a comprehensive set of information about the transaction, the customer and the teller requesting the override.

A supervisor with a queue of override requests must approve or deny the requests one at a time. Supervisor overrides are recorded in your journal.

### 3.8 Performing a Local Supervisor Override

Your supervisor can perform the override at your workstation.

To perform a local supervisor override

1. Complete the Transaction details. The fields are described in the following table.

Field	Comments	
Teller Name	Displays the name of the teller who made the override request.	
Teller Branch	Displays the branch code.	
Transaction Type	Displays the type of transaction.	
Referral Id	Displays an ID for the override request.	
Transaction Currency	Displays the currency of the transaction.	
Transaction Amount	Displays the amount of the transaction.	
Fee Currency	Displays the currency in which the fee for the transaction is charged.	
Fee Amount	Displays the fee for the transaction.	
Reason for Referral Displays the reason for the override request, for example, "The usentitlements limit has been exceeded".		
Reason for Decline	Supervisor to type a reason for declining the override request.	
Check to Approve	Supervisor to select this, if the request is to be approved.	

2. A message is displayed informing you that you have exceeded your limit. Click Supervisor override locally. Otherwise, wait until the Supervisor Override screen is displayed.

- Allow your supervisor to type the user ID and password to authorize, and to click Approve or Decline.
- **4.** Click Submit.

### 3.9 Performing a Remote Supervisor Override

If you are a supervisor, a list of override requests is displayed on your workstation, and you can accept or decline the override request at your workstation.

To perform a remote supervisor override

- 1. Click the flashing indicator icon present in the status line at the bottom right of the screen. The Retrieve Supervisor Referrals screen is displayed.
- Details of Override Reasons section present towards the end of the screen have the following fields.

Field	Comments
Reason for Referral	Displays the reason for the override request, for example, "The users entitlements limit has been exceeded".
Reason for Decline	Supervisor to type a reason for declining the override request.
Check to Approve	Supervisor to select this, if the request is to be approved.

- **3.** Click Approve or Decline.
- Click Submit.

Siebel Branch Teller sends a response to the teller.

### 3.10 Temporary Logoff

The Temporary Logoff operations performed to temporarily disconnect from Siebel Branch Teller, leaving the workstation available for another user. You do not have to perform a Teller End of Day operation at this time, but the cashbox remains assigned to you, and the cash totals are not affected. For more information, see, Section 4.3, "Logging Out Temporarily".

### 3.11 Logging Off

The Logoff operation is performed to log out of Siebel Branch Teller. If a Teller End Of Day operation is required, a message is displayed to prompt you to perform that operation. After you log out, Siebel Branch Teller closes. For more information, see, Section 8.15, "Logging Out".

# **Searching Customer Details and other** commonly used features

There are some commonly used features in Siebel Branch Teller.

This chapter describes the commonly used features that are listed in the top left side menu that appear as static menu, always visible to you. It also describes details about the Customer relationship Console. Following are the topics:

- Section 4.1, "Searching Customer Information"
- Section 4.2, "Incomplete Customer Sessions"
- Section 4.3, "Logging Out Temporarily"
- Section 4.4, "The Customer Relationship Console"

## 4.1 Searching Customer Information

You use the Customer Search operation to search for details about customers. You can use account details found in the search when you perform financial transactions.

#### To search for customer information

- Navigate to the Customer Search screen (Fast Tran 11).
- Enter the search details in one of the following ways:
  - Click Branch Code and type a branch code and account number.
  - Click First Name, and then type the first and second name of the customer. You must type the names exactly as they are spelled, including any hyphens or apostrophes. Each name must have a capital letter as the first letter. These fields are case sensitive.
  - Click Card Swipe, and then swipe the customer's card.
  - Click Credit Card, and then type the card number without spaces, and the expiration date.
- **3.** Click Search. If more than one customer with the same name is found, you are prompted to choose the customer that you require.

The customer information is displayed, including the customer relationship console (CRC). For more information about the CRC and the information it provides, see Section 4.4, "The Customer Relationship Console".

### 4.2 Incomplete Customer Sessions

When a customer session is paused unintentionally or intentionally, it becomes an incomplete customer session. Examples of unintentional cases are user session timeout and client system failure. Intentional cases include suspending a customer session temporarily and completing it sometime later as planned. Sessions transferred from Cashless User (Enable Cash Transactions) also will be displayed in Incomplete Sessions. When such sessions are transferred to a specific teller, an alert is sent to the target teller with the menu item 'Incomplete Customer Sessions' blinking in red.

After you log in, Siebel Branch Teller checks if there are any incomplete sessions. If there are incomplete sessions, a message is displayed on the homepage alerting you that you have incomplete sessions. Note that, the tellers can view and complete only their incomplete sessions, whereas supervisors can view and complete their own and also other tellers' incomplete sessions.

#### To view incomplete customer sessions

- Navigate to Incomplete Sessions (Fast Tran 12). The incomplete sessions in Active or Suspended or Transferred status are displayed here.
- To resume an incomplete session, select the session and click the Restart Session button.

It is not possible to perform a Teller End of Business Day or Teller End if you have incomplete sessions. You will be prompted to complete the incomplete sessions.

## 4.3 Logging Out Temporarily

You use the Temporary Logoff operation to temporarily disconnect from Siebel Branch Teller, leaving the workstation available for another user. You can use the operation if you are a teller, supervisor, vault administrator, or customer service representative.

You do not have to perform a Teller End of Day operation at this time, but the cashbox remains assigned to you, and the cash totals are not affected.

#### To log out temporarily

- Navigate to Temporary Logoff screen (Fast Tran 13).
- Click OK on the Temporary Logoff screen.

### 4.4 The Customer Relationship Console

The CRC displays Summary and Account information for a customer, and when the Siebel Branch Teller application is integrated with the Siebel Financial Services 8.x application, the CRC displays alerts and offers that you can extend to the customer. The CRC provides information on separate tabs: Summary, Accounts, Offers, Alerts, and SRs. You display the different types of information by clicking on the corresponding tab, or by pressing Alt together with the underlined character on the

You can click Contacts at the top right of the CRC to return to the Summary view from anywhere in the CRC.

If they are available from the Siebel application, the customer's photograph is displayed together with the customer's signature at the left side of the CRC. Click Signature to view the signature.

You can select an offer available to the customer and create a referral for that offer. You can click My Referrals at the top right of the CRC to display a list of your referrals.

**Note:** You can click the customer-centric/account-centric indicator on the status bar at the bottom of the screen depending on whether you want to receive customer details. When the red stop-sign is shown, the CRC is not displayed.

### 4.4.1 Viewing Customer Summary Information

When you have performed a customer search, the CRC is displayed. The full name of the customer is displayed in the upper-left corner. The authentication information, if available, is displayed under the name. The Summary view of the CRC provides an overview of the following customer information:

- **Personal.** Address and other contact information
- **Customer ID.** Partially hidden for security reasons
- **Branch information.** The customer's home branch
- Coverage. The customer's personal financial advisor
- **Financial profile information.** The customer's total assets and liabilities
- **Customer value.** Given as a number of gold stars (the greater the number of stars, the more valuable the customer)
- Offers. The offers currently available to the customer
- **Alerts.** The alerts currently applying to the customer
- **SRs.** The service requests relevant to the user.

Click on an entry in the list of offers to go to the Offers view, which displays detailed information about the offer and allows you to make a referral.

Click on an entry in the list of alerts to go to the Alerts view, which displays detailed information about the alert.

### 4.4.2 Viewing Customer Account Information

The Accounts view of the CRC displays key information about the customer's accounts, see the following table.

Customer Account Information in the CRC

Field	Comment	
Account Number	Displays the account number for authentication purposes.	
Account Type	Displays the product name.	
Joint Account	Indicates whether the account is a joint account.	
Account Name	Displays the name of the account holder, or account holders in the case of a joint account.	
Currency	Displays the currency.	
Current Balance	Displays the current balance of the account (including deposited funds not yet posted).	
Available Balance	Displays the available balance of the account.	
Status	Displays the status of the account.	

When you select an account from the list, you can go to Customer Transactions, Customer Services, or Customer Inquiries operation, and the account you selected is displayed in the Account Details section for the operation.

### 4.4.3 Viewing Customer Offer Information

The Offers view of the CRC displays information about offers that your organization wants to extend to the customer. You can record the customer's response to the offer. For customers who accept an offer you create a referral.

The fields in the list of offers are shown in the following table:

Customer Offer Information in the CRC

Field	Comment
Offer Name	Displays the name of the offer.
Product Name	Displays the name of the product to which the offer applies.
Expiration Date	Displays the date by which the offer ends.
Description	Displays a description of the offer.

To display more detail about an offer, select the offer from the list. More detail about the offer is displayed, and you can create a referral for the offer.

### 4.4.4 Creating a Referral

When you make customers aware of an offer, they can accept or reject the offer. When they accept the offer, you can create a referral for the offer. A referral captures information about the offer that is useful in customer relationship management.

Click My Referrals, on the top right of the CRC, to display a list of referrals that you have created.

#### To create a referral

- **1.** Select the Offers view in the CRC.
- Select the relevant offer from the list.
- Select Accept in the Response drop-down list.
- Click Next.
- Complete a New Referral record. Some fields are described in the following table.

Field	Comment	
Referral Name	Displays a name that uniquely identifies the referral record. The name includes the customer's name and the date and time.	
Comments	Type a comment to explain the context of the referral, if required.	
Product	Displays the name of the product to which the offer applies.	
Contact	Displays the customer's name.	
Sales Method	Select the sales method used to solicit the new referral.	
Company	Displays the name of the company that made the offer.	
Source	Displays the source of the referral; that is, Teller.	
Immediate Attention	Click if the referral requires immediate attention.	

- **6.** Click Save.
- **7.** If the information for the referral is complete, click Accept. Otherwise, click Previous if you want to change anything.

The referral record is created, and the offer is removed from the list. You can view this record and your other referrals by clicking My Referrals.

### 4.4.5 Viewing Customer Alert Information

The Alerts view of the CRC displays information about the customer that has been captured throughout your organization. Alerts contain information that you might have to act upon, or inform the customer about. The fields in the list of alerts are listed in the following table.

Customer Alert Information in the CRC

Field	Comment	
Message	Displays the description of the alert	
Priority	Displays the relative importance of the alert	
Туре	Displays information, for example, about which person in your organization generated the alert	
Start Date	Displays the date on which the alert was created	

### 4.4.6 Viewing Service Request Information

The SRs view of the CRC displays information about service requests (SRs) relevant to the customer. The list of service requests is provided for your information only. The fields in the list of SRs are listed in the following table.

Service Request Information in the CRC

Field	Comment	
SR#	Displays an identifier for the service request.	
Open Date	Displays the date when the SR was created.	
Due Date	Displays the date when the SR is due to be completed.	
Туре	Displays the type of the SR.	
Area	Displays the area affected by the SR.	
Status	Displays the status of the SR, for example, Open, Resolved.	
Sub-status	Displays further status information.	
Owner	Displays the name of the owner of the SR.	

The Customer Relationship Conso	The	Customer	Relationship	Console
---------------------------------	-----	----------	--------------	---------

# **Performing Customer Financial Transactions**

You use the Customer Transactions to perform financial transactions for the customers. When you select one of the operations, a customer session starts, and you can then perform a number of transactions in the session.

If the amount of a transaction exceeds your authorized limit, your supervisor must override the limit.

When you complete a transaction, Siebel Branch Teller creates a journal entry, and updates the cashbox, account balance, and check batch, if applicable. A customer receipt is printed at the end of the customer session.

This chapter covers the financial transactions that you can perform. It includes the following topics:

- Section 5.1, "Customer Sessions"
- Section 5.2, "Currency Transaction Reporting"
- Section 5.3, "Common Tasks for Financial Transaction Operations"
- Section 5.4, "Making a Deposit"
- Section 5.5, "Making a Withdrawal"
- Section 5.6, "Cashing a Check"
- Section 5.7, "Transferring Funds Between a Customer's Accounts"
- Section 5.8, "Transferring Funds to Another Account"
- Section 5.9, "Making a Wire Transfer"
- Section 5.10, "Making International Transfer"
- Section 5.11, "Making a Payment to a Utility"
- Section 5.12, "Making a Cash Denomination Exchange"
- Section 5.13, "Making a Foreign Currency Exchange"
- Section 5.14, "Issuing a Local Draft"
- Section 5.15, "Issuing a Draft in a Foreign Currency"
- Section 5.16, "Issuing a Cashier's Check"
- Section 5.17, "Issuing a Money Order"
- Section 5.18, "Purchasing U.S. Savings Bonds"
- Section 5.19, "Making U.S. Savings Bond Inquiries and Redeeming Bonds"

### 5.1 Customer Sessions

When you start a financial transaction, a customer session starts, and you can then perform one or more transactions within the session. During the session, a transaction register is displayed.

The transaction register contains various indicators and buttons, and a list of transactions that you have performed during the customer session. The indicators and buttons are described in the following table.

Fields and Controls in the Transaction Register

Field	Comment
Currency	Displays the currency of the customer session. By default the base currency of the bank is displayed. You can select other currencies from this drop-down list; the Pay To and Get From amounts then change accordingly.
Pay To	Displays the amount to be paid to the customer at the end of the session.
Get From	Displays the amount to be obtained from the customer at the end of the session.
Transfer	For the Cashless User (Enable Cash Transactions). Click to transfer the customer session to another teller.
End	Click to end the customer session. Siebel Branch Teller prints a combined receipt with all transactions at the end of the session.
Suspend	Click to suspend the customer session that you plan to resume later. After ending the Customer Session, you are directed to the Home page. To resume the suspended sessions, go to Incomplete Sessions.
Reverse	Click to reverse one or more transactions selected in the list.
Cancel Session	Click to cancel the customer session. If you have posted any transactions in the session, they are reversed.

You can cancel, reverse, or suspend the individual transactions within a customer session. In addition, you can cancel, reverse, or suspend the entire customer session. If you cancel the customer session, all transactions completed during the session are reversed.

The list of transactions includes the columns shown in the following table.

Columns in the List of Transactions

Field	Comment
Date/Time	Displays the date and time of the transaction.
Txn Type	Displays the transaction type. If the transaction has been reversed, the word "Reversed" is displayed after the transaction type.
Txn Status	The status of the transactions, for example, cleared, pending.
Source Account	The source account associated with the transaction (where relevant).
Destination Account	The destination account associated with the transaction (where relevant).
Currency	The currency of the transaction.
Amount	The amount of the transaction.

At the end of a customer session, if the amount in the Pay To or Get From field is greater than 0.00, the Cash In and Cash Out Details screen is displayed, and you must enter the denominations of the cash to be obtained from the customer or paid to the customer.

#### To complete the Cash In and Cash Out details

- If there is cash to be obtained from the customer, the Total Amount field in the Cash In Details panel displays the appropriate amount; type the quantity of each type of note and coin obtained from the customer.
- If there is cash to be paid to the customer, the Total Amount field in the Cash Out Details panel displays the appropriate amount; type the quantity of each type of note and coin paid to the customer.
- Click Submit.
- Repeat steps 1 through 3 for each of the currencies used in the customer session.

If currency transaction reporting is necessary, the Currency Transaction Report screen is displayed, and you have to complete a CTR form. For more information, see Section 5.2, "Currency Transaction Reporting".

**Note:** Cashless User (Enable Cash Transactions) cannot complete a Customer Session. These sessions can be transferred to another specific user who has a cashbox or to all users in the branch with cashboxes. Such Customer Sessions are called Transferred Session and considered as one of the types of **Incomplete Customer Sessions**.

### 5.1.1 Reversing Transactions

Sometimes you need to reverse a transaction, for example, if you make an error in entering the transaction details. You can reverse a transaction by:

- Using the Reverse button in a customer session.
- Using the Reverse button when viewing your journal, if you already submitted the transaction; see Section 8.5.1, "Displaying the Journal, Reversing Transactions, and Printing a session receipt".

When you reverse a transaction that involved a negotiable instrument such as a deposit, payment, bond purchase, or cash check, the check batch is updated to decrease the batch amount and the number of items in the batch.

When you reverse a cash transaction (cash deposit, withdrawal), the current and available balances of the destination account are revised. When you perform an inter-account transfer the customer current and available balances of the source and destination account are revised.

When you reverse a transaction involving an on-us check:

- For deposits, the current balance and the available balances of the source and destination account are revised.
- For cash check, payment, and bond purchase transactions, the current balance and the available balances of the source account are revised.

For a transaction involving an "other" check, the current balance of the destination account is revised.

## 5.2 Currency Transaction Reporting

To comply with Bank Secrecy Act and USA Patriot Act requirements, your bank must file currency transaction reports (CTR) with the Financial Crimes Enforcement Network (FinCen). You must complete FinCen form 104 for every transaction involving cash-in or cash-out amounts exceeding \$10,000, unless the relevant customer or the account are exempt from currency transaction reporting.

A bank has the right to exempt certain qualifying customers from the CTR process as long as these customers are first registered with FinCEN. The list of exemptions is maintained using the Administration Console, see Siebel Retail Finance Banking Application Administration Guide for more information.

At the end of a customer session, Siebel Branch Teller checks whether CTR processing is necessary. If the cash-in amount or the cash-out amount for a transaction is greater than \$10,000, and there is no exemption, then CTR processing is necessary. For CTR processing, multiple transactions are treated as single transactions if they are by, or on behalf of, the same person. In calculating the cash-in and cash-out amounts, Siebel Branch Teller converts amounts in currencies other than U.S. dollars into dollar equivalents.

If the cash-in and cash-out amounts both exceed \$10,000, you only need to complete one CTR.

When CTR processing is necessary, you are presented with a FinCen form 104 with most of the information already filled in. If all required information is not available when you complete the form, you can suspend CTR processing and resume processing later. However, all CTR forms must be submitted either individually or batched to FinCEN's processing center within 15 days. Your bank is also responsible for filing CTR forms.

If there were more than three customers involved in the customer session, you must process the form manually.

### 5.2.1 Completing the Currency Transaction Report

When a customer session ends, and currency transaction reporting is required, a CTR template screen is displayed. The CTR form contains the following sections:

- **PART 1 Person(s) Involved in Transaction(s).** This part contains two sections.
  - Part 1 Section A. This section contains information relating to the person on whose behalf the transaction is being conducted. If an individual conducts a transaction on his or her own behalf, and on behalf of another person, you must complete Section A for each person on whose behalf the transaction is being conducted.
  - **Part 1 Section B.** This section contains information relating to a person who is conducting a transaction on behalf of another. If an individual conducts a transaction on his or her own behalf, you must leave Section B blank.
- **PART 2 Amount and Type of Transaction(s).** This part identifies the type of transactions and the amount involved.
- **PART 3 Financial Institution Where Transaction(s) Take Place.** This part contains details about your bank.

Your bank stores and regularly updates the correct FinCen form and instructions for completing the form.

To complete a currency transaction report by the branch supervisor

1. Complete the fields of the form that are not already completed; most of part 1 section A and part 2 are filled in automatically.

Make sure that you enter the minimum information that is required. These fields are:

Field	Comment
Part 1, Section A	
Issued By	Type the state that issued the means of identification.
Part 1, Section B (if required)	
Last Name	Type the name of the person or organization.
First Name	Type the first name of the person, or leave blank for an organization.
Address	Type the address of the person or organization.
City	Type the city.
Zip Code	Type the zip code.
Date of Birth	Type the date of birth, or select it from the popup.
If an individual, method of ID	Select the checkbox corresponding to the means of identification.
Issued By	Type the state that issued the means of identification.
Part 3	
Name of Financial Transaction	Type the full legal name of your financial institution.
Examiner Code Number	Type the code of the regulator or BSA examiner. See the instructions supplied with form 104 for the list of codes,
Address	Type the address of your financial organization.
SSN	Type the social security number of your financial institutions principal owner.
City	Type the city.
State	Type the state.
Zip Code	Type the zip code.
Routing MCR Number	Type the routing Magnetic Ink Character Recognition (MICR) number.

- 2. If there is more than one person and up to three persons involved, click Multiple Persons, and enter the information for the additional persons.
- If you do not have all of the information to complete the form, click Suspend.
- Click Submit.
- To print the filled CTR form, click View Report button.

### 5.2.2 Resuming a Suspended Currency Transactions Report

You can retrieve suspended CTR forms in two ways:

By clicking the CTR icon on the status line. At each start of day, Siebel Branch Teller checks for suspended CTR forms. If a CTR form is not completed in 3 working days, the status remains suspended, but the icon for the CTR form is no longer displayed on your status line. Instead, an icon is displayed on the status line for the branch administrator.

By searching for CTR transactions in the journal. For more information, see Section 8.5.1, "Displaying the Journal, Reversing Transactions, and Printing a session receipt".

In both cases, the Journal Summary screen is displayed and you can select a suspended CTR. When you click Details, the suspended CTR form is retrieved and you can continue with completing the form.

### 5.2.3 Printing Currency Transaction Report Form

There are specified amounts that you can deposit or withdraw without printing a Currency Transaction Report (CTR) form. The currency transactions like deposit or withdrawal of amounts more than the specified amount in a single or multiple transactions requires you to print CTR form. The CTR form is generated using a tool called Oracle Business Intelligence Publisher (Oracle BI Publisher).

#### To print a CTR report after deposit

- Make cash deposit or withdrawal of more than the specified amount in a single or multiple transactions. See Section 5.4, "Making a Deposit" and Section 5.5, "Making a Withdrawal".
- **2.** After the transactions are complete, click End. The Cash In Details / Cash Out Details window appears.
- **3.** Enter the cash denominations and click Submit. CTR Details window appears.
- Enter SSN number. Click Submit. Currency Transaction Report form appears.
- 5. Enter the details and fill this form. Click Submit. The Submit button gets disabled and the Print button gets enabled.
- Click View Report. The CTR form appears in the Internet Browser. You can save this form as a pdf or take its print out. In Siebel Branch Teller, you are taken to the Home page.

### 5.3 Common Tasks for Financial Transaction Operations

Some tasks are common to most of the financial transaction operations:

- Section 5.3.1, "Selecting the Account"
- Section 5.3.2, "Entering Check Details"
- Section 5.3.3, "Entering Payment Method Details"
- Section 5.3.4, "Waiving the Fee for a Transaction"
- Section 5.3.5, "Entering Customer Authentication Details"
- Section 5.3.6, "Receipt Printing Transactions"

These tasks are described in the following sections.

### 5.3.1 Selecting the Account

Some transactions require the branch code and account number details for an account-holding customer. You can provide these details in any of the following ways:

- In account-centric mode, you can type the branch code and account number of the customer's account, and click Search.
- In offline mode, you must type the branch code and account number and click Search.
- In customer-centric mode, you can first perform a Customer Search operation (Fast Tran 11). When you then start a transaction, the branch code and account number are automatically displayed in the Account Details field.

### 5.3.2 Entering Check Details

A number of financial transactions require that you enter details of checks. If the facility is available, you can use a MICR device to swipe checks to obtain the MICR line information such as ABA number (branch code), account number, and check sequence number details.

#### To enter check details

- Click Check Details on the transaction screen.
- If you have a MICR reader, swipe the checks.
- If you do not have a MICR reader, complete the Check Details manually.

Field	Comment
Branch Code	Type the ABA number (branch code) on the check.
Account Number	Type the account number on the check.
Check Number	Type the number of the check.
Amount	Type the amount of the check.
Document Type	Select On-us if the check was issued by your bank, or Other, if the check was issued by another bank.

**4.** Click Submit when you have entered all the details.

### 5.3.3 Entering Payment Method Details

A number of financial transactions require a payment from the customer. The following methods can be used for payments:

- Cash
- Check
- Check and cash
- Funds transfer (account-holding customers only)

The payment amount from the customer can be more than the amount of the current transaction. This allows you to perform multiple transactions with one payment. If the payment amount is greater than the amount of the transaction (including any fee) the virtual cash amount is updated.

To facilitate over-payment by the customer and facilitate providing cash back to the customer, the user can enter an amount above that required to balance a transaction.

#### To enter payment method details

**1.** Navigate to the relevant transaction screen.

2. Complete the Payment Method details. The fields are described in the following

Field	Comment
Total Amount	Displays the total amount of the payment, including any fee.
Cash Amount	Type the amount of cash for the payment.
Total Check Amount	Type the amount in checks for the payment.
Number of Checks	Type the number of checks to be used for payment. Click Check Details to enter the details of the checks.
Funds Transfer Amount	Type the amount of funds to be transferred for the payment.

3. Click Submit.

### 5.3.4 Waiving the Fee for a Transaction

Fees are applied to a number of financial transactions, and the fee is calculated based on the transaction amount, account type, customer type, and transaction type. However, you can waive the fee for a transaction if required.

### To waive the fee for a transaction

- In the relevant transaction screen, type the amount of the transaction.
- Complete the Fee Details. Some fields are described in the following table.

Field	Comment
Fee Amount	Displays the fee for the transaction.
Waive	Select this radio button to waive the transaction fee.
Reason	Select a reason for waiving the fee.

### 5.3.5 Entering Customer Authentication Details

A number of transactions might, depending on your bank's policy, require you to authenticate the customer's identity. You can confirm identity by checking a passport, social security number, or by using whatever document is acceptable at your bank.

When you have entered authentication details, you do not have to enter them again during the customer session. The details are not displayed after you enter them.

#### To authenticate customer details

- Select the authentication method from the Authenticate Customer drop-down list.
- Type the ID Number for the ID being used in the authentication.

Siebel Branch Teller records the customer authentication details.

## 5.3.6 Receipt Printing Transactions

The receipts that are printed adhere to a standard format. However, the receipt printing format varies based on each transaction. Following are the transactions for which receipts are printed.

Withdrawal

- Deposit
- **Funds Transfer**
- Inter Account Transfer
- **Payment**
- Local Draft
- Foreign Deposit
- Foreign Withdrawal
- Bond Redemption
- Foreign Draft
- Cash Exchange
- Foreign Exchange
- General Ledger
- International Transfer
- Wire Transfer

# 5.4 Making a Deposit

You use the Deposit operation to deposit cash and checks to a customer's account. You can make a deposit in the base currency for your bank or in any currency that is held in your cashbox. If the customer wants to deposit money in more than one currency, you must perform a separate transaction for each currency.

Depending on the rules that apply at your bank, a customer might be able to make a partial deposit by depositing a check and receive part of the value of the check back in cash.

If the facility is enabled at your bank, you can record the denominations of cash payments. This is designed to make cash management in your branch easier.

### To make a deposit

- Navigate to the Customer Transactions > Deposit screen (Fast Tran 21).
- Enter the account details if they are not already displayed. See Section 5.3.1, "Selecting the Account".
- Complete the Deposit Details. The fields are described in the following table.

Field	Comment
Deposit Currency	Select the currency of the deposit. This field defaults to the base currency of your bank.
Cash Amount	Type the amount of cash to be deposited.
Total Check Amount	Type the total for the checks to be deposited, and click Check Details. Enter the details for each check. See Section 5.3.2, "Entering Check Details".
Number of Checks	Type the number of checks.
Cash Back Amount	Type the amount of cash to be given back to the customer, if partial deposits are possible.

Field	Comment
Deposit Amount	Displays the total amount deposited. This is the total amount of cash plus checks, minus any cash back amount.
Rate	Displays the exchange rate for a foreign currency deposit.
Commission	Displays the commission applicable for a foreign currency deposit.
Total Deposit Amount	Displays the total amount of the deposit in the base currency. For a foreign currency deposit, this is calculated from the Total Deposit Amount, Rate, and Commission values.

- **4.** Enter the authentication details for the customer, if required; see Section 5.3.5, "Entering Customer Authentication Details".
- 5. Click Submit.

The transaction register and journal are updated with details of the transaction.

### 5.5 Making a Withdrawal

You use the Withdrawal operation to withdraw money for a customer. You can make a withdrawal in the base currency for your bank or in any currency that is held in your cashbox. If the customer wants to withdraw money in more than one currency, you must perform a separate transaction for each currency.

If the facility is enabled at your bank, you can record the denominations of cash withdrawals. This is designed to make cash management in your branch easier.

#### To make a withdrawal

- Navigate to the Customer Transactions > Withdrawal screen (Fast Tran 22).
- Enter the account details if they are not already displayed; see Section 5.3.1, "Selecting the Account".
- Complete the Withdrawal Details. The fields are described in the following table.

Field	Comment
Withdrawal Currency	Select the currency of the withdrawal. This field defaults to the base currency of your bank.
Cash Amount	Type the amount of cash to be withdrawn.
Rate	Displays the exchange rate for a foreign currency withdrawal.
Commission	Displays the commission applicable for a foreign currency withdrawal.
Withdrawal Amount	Displays the total amount of the withdrawal in the base currency. For a foreign currency withdrawal, this is calculated from the Cash Amount, Rate, and Commission values.

- Enter the authentication details for the customer, if required; see Section 5.3.5, "Entering Customer Authentication Details".
- Click Submit.

### 5.6 Cashing a Check

You use the Cash Check operation to cash a check for a customer. A check can be from your bank referred to as "on us", or from another bank, referred to as "other".

There must be sufficient funds in the customer's account to cover the check. A customer who is not an account holder is charged a fee; you can waive this fee, if required.

The source of the funds for an "on-us" check is the account that the check is drawn on. If the funds are available, they are withdrawn from that account. If the funds are not available, a message is displayed and the transaction is cancelled.

The source of the funds for an "other" check is an internal suspense account. If the person cashing the check is not a customer, you require a supervisor override to complete the transaction.

#### To cash a check

- Navigate to the Customer Transactions > Cash Check screen (Fast Tran 23).
- Enter the account details if they are not already displayed; see Section 5.3.1, "Selecting the Account".
- Complete the Payment Method Details. The fields are described in the following table.

Field	Comment
Total Check Amount	Type the total for the checks to be cashed.
Number of Checks	Type the number of checks and click Check Details. Enter the details for the checks. See, Section 5.3.2, "Entering Check Details".

After you complete the Payment Method details, the Cash Back To Customer field shows the amount due to the customer.

- **4.** If there is a fee, waive the fee if required; see Section 5.3.4, "Waiving the Fee for a Transaction".
- Enter the authentication details for the customer, if required; see Section 5.3.5, "Entering Customer Authentication Details".
- Click Submit.

### 5.7 Transferring Funds Between a Customer's Accounts

You use the Inter-Account Transfer operation to transfer funds to another account held by a customer at your bank. Inter account transfer can be done only in a Customer Session.

#### To transfer money to another account

- Perform a Customer Search.
- Navigate to the Customer Transactions > Inter-Account Transfer screen (Fast Tran
- Enter the destinations account details if they are not already displayed:

Field	Comment
To Account Number	Select the account number of the destination account.
To Branch Code	Displays the branch code of the destination account.
Amount	Type the amount to be transferred.
Date	Displays the date on which the transfer will take place.

- **4.** Enter the authentication details for the customer, if required; seeSection 5.3.5, "Entering Customer Authentication Details".
- **5.** Click Submit.

## 5.8 Transferring Funds to Another Account

You use the Funds Transfer operation to transfer money to an account held by another customer within your bank or to an account held by a customer in a different bank.

#### To transfer funds

- Navigate to the Customer Transactions > Funds Transfer screen (Fast Tran 25).
- Enter the account details if they are not already displayed. See, Section 5.3.1, "Selecting the Account".
- Complete the Transfer Details. Some fields are described in the following table.

Field	Comment
Beneficiary Name	Select the name of the beneficiary for the transfer, or select New if there are no beneficiaries. A beneficiary is a person or company to whom some form of payment is made from a customer's account. If necessary, you can click Maintain Beneficiaries to maintain the details of beneficiaries.
Amount	Type the amount to be transferred.
Effective Date	Type the date that the transfer will take place, if it is not the current processing date.
Account Number	Type the beneficiary's account number, if this is a new beneficiary.
Narrative	Type a description, if required.

- **4.** If required, waive the transaction fee. See, Section 5.3.4, "Waiving the Fee for a Transaction".
- Enter the authentication details for the customer, if required. See, Section 5.3.5, "Entering Customer Authentication Details".
- 6. Click Submit.

### 5.9 Making a Wire Transfer

You use the Wire Transfer operation to transfer money from an account held by a customer within your bank to an account held by a customer in a different bank in same country or different country. The transfer operation can also be performed by a non-customer (non-existing customer) by mode of cash.

If you are transferring within a country, it is a domestic transfer and for that, you need to know the American Bankers Association (ABA) routing number for filling the Beneficiary details. ABA routing number is usually the nine digit routing transit number that appears on the bottom of negotiable instruments such as checks identifying the financial institution on which it was drawn.

If you are transferring to another country, you need to know the International Bank Account Number (IBAN) for filling the Beneficiary details. IBAN is an international standard for identifying bank accounts internationally.

To make a wire transfer of funds

- 1. Navigate to the Customer Transactions > Wire Transfer screen (Fast Tran 26).
- Select whether you need to make a domestic transfer or international transfer.
- Select the Source of funds based on whether you need to make the transfer through the Account or through Cash. Enter the account details if you had selected Account.
- Complete the Transfer Details. The required fields are described in the following

Field	Comment
Source Amount	Type the amount to be transferred.
Currency	Set to account currency in case of account transfer or base currency in case of cash
Processing Date	Displays the processing date. If your branch is operating in AM/PM processing mode, this date is different from the calendar date in the P.M. period.
Beneficiary Currency	Select the beneficiary currency. This field defaults to the base currency of your bank.

- Complete the Beneficiary Details.
- The fees can paid either through account or by cash. If required, waive the transaction fee. See, Section 5.3.4, "Waiving the Fee for a Transaction".
- Extended Authenticate details needs to be filled in, when cash transfer is performed. It won't appear for an account transfer with customer session.
- Enter the authentication details for the customer, if required. See, Section 5.3.5, "Entering Customer Authentication Details".
- Click Submit.

# 5.10 Making International Transfer

You use the Wire Transfer operation to transfer money from an account held by a customer within your bank to an account held by a customer in a different bank in different country. You need to know the International Bank Account Number (IBAN) for filling the Beneficiary details. IBAN is an international standard for identifying bank accounts internationally.

#### To make an international transfer of funds

- Navigate to the Customer Transactions > International Transfer screen (Fast Tran
- Select the Source of funds based on whether you need to make the transfer through the Account or through Cash. Enter the account details if you had selected
- Complete the Transfer Details. The fields are described in the following table.

Field	Comment
Source Amount	Type the amount to be transferred.
Currency	Set to account currency in case of account transfer or base currency in case of cash

Field	Comment
Processing Date	Displays the processing date. If your branch is operating in AM/PM processing mode, this date is different from the calendar date in the P.M. period.
Beneficiary Currency	Select the beneficiary currency. This field defaults to the base currency of your bank.

- Complete the Beneficiary Details.
- The fees can paid either through account or by cash. If required, waive the transaction fee. See, Section 5.3.4, "Waiving the Fee for a Transaction".
- **6.** Extended Authenticate details needs to be filled in, when cash transfer is performed. It won't appear for an account transfer with customer session.
- **7.** Enter the authentication details for the customer, if required. See, Section 5.3.5, "Entering Customer Authentication Details".
- Click Submit.

## 5.11 Making a Payment to a Utility

You use the Payments operation to pay a utility bill for a customer.

#### To pay a utility bill

- 1. Navigate to the Customer Transactions > Bill Payment screen (Fast Tran 28).
- Enter the account details if they are not already displayed; see Section 5.3.1, "Selecting the Account".
- Complete the Utility Payment Details. Some fields are described in the following table.

Field	Comment
Utility Name	Select the name of the utility to be paid.
Reference Number	Displays the reference number of the utility payment.
Branch Code	Displays the utility's branch code.
Account Number	Displays the utility's account number.
Effective Date	Type the date that the payment will take place for funds transfer, if it is not the current processing date.
Amount	Type the amount to be paid.
Narrative	Type a description, if required.

- If required, waive the transaction fee; see Section 5.3.4, "Waiving the Fee for a Transaction".
- Enter the appropriate Payment Method details; see Section 5.3.3, "Entering Payment Method Details".
- Enter the authentication details for the customer, if required; see Section 5.3.5, "Entering Customer Authentication Details".
- 7. Click Submit.

## 5.12 Making a Cash Denomination Exchange

This operation provides functionality for the bank to change denominations of one currency for other denominations of the same currency, for a customer or non-customer of the bank. The application manages the totals for each denomination in the cash drawer, therefore this operation is required to update the cash drawer denomination amounts after a cash denomination exchange. This transaction is recorded in the Teller's electronic journal.

#### To make a cash denomination exchange

- Navigate to the Customer Transactions > Cash Exchange screen (Fast Tran 29).
- Select the cash exchange currency and enter the amount.
- Click Submit. 3.
- Enter the denomination amounts for both the Cash In Details and Cash Out Details screens.
- 5. Click Submit.

A receipt with transaction details is printed once the transaction has completed.

## 5.13 Making a Foreign Currency Exchange

This operation provides the facility to exchange cash in one currency for cash in another currency. The user may also exchange foreign currency checks for cash in the home currency. This Foreign Exchange operation is used for the following types of foreign currency exchange transactions:

- Capture and calculate the buying and selling of foreign currency cash.
- Capture and calculate cross currency exchanges.
- Capture and calculate the buying of foreign currency checks.

The denomination amounts are updated in the system for the Teller's cashbox and the transaction is recorded in the Teller's electronic journal.

#### To make a foreign currency exchange

- Navigate to the Customer Transactions > Foreign Exchange screen (Fast Tran 210).
- Select the Instruction Type, for example, Buy Cash, Buy Check, Cross Exchange, or Sell Cash.
- **3.** Select the foreign currency that the customer wishes to buy, sell, or cross exchange.
- Enter the currency amount.
- Click Submit.

A receipt with transaction details is printed once the transaction has completed.

## 5.14 Issuing a Local Draft

You use the Local Draft operation to issue a draft on behalf of a customer. The payment method is by cash, check, or funds transfer from the customer's account.

Before you start, you must know the serial number of the draft.

#### To issue a draft

Navigate to the Customer Transactions > Local Draft screen (Fast Tran 211).

2. Complete the Draft Payment Details. The required fields are described in the following table.

Field	Comment
Issue Dynamically Allocated Serial Number	Select this radio button to automatically assign a serial number to the draft.
Serial Number	Select the radio button and type the serial number of the draft.
Payee Name	Type the name of the payee.
Amount	Type the amount of the draft.
Date	Displays the date.
Memo	Type a memo, if required.

The Purchaser details contain the customer's details.

- If required, waive the transaction fee; see Section 5.3.4, "Waiving the Fee for a Transaction".
- Enter the appropriate Payment Method details; see Section 5.3.3, "Entering Payment Method Details".
- Click Submit on the Issue Draft screen.

# 5.15 Issuing a Draft in a Foreign Currency

You use the Foreign Draft operation to issue a draft in a foreign currency on behalf of a customer. The payment method is by cash, check, or funds transfer from the customer's account.

Before you start, you must know the serial number of the draft.

#### To issue a foreign draft

- Navigate to the Customer Transactions > Foreign Draft screen (Fast Tran 212).
- Select the currency for the draft from the FX Currency drop-down list.
- Click FX Draft Details, and complete the draft details as described in the following table.

Field	Comment
Draft Number	Type the number of the draft.
Payee Name	Type the name of the payee.
FX Currency	Displays the currency of the draft.
FX Draft Amount	Type the amount of the draft.
Date	Displays the date.

### 4. Click Submit.

You are returned to the FX Issue Draft screen, and the amount of the draft is now displayed in the FX Draft Amount field.

Complete the remaining Draft Payment Details by clicking the Calculate button. The fields are described in the following table.

Field	Comment
Base Account Currency	Displays the base currency of the account.
Rate	Displays the exchange rate. The Rate, Commission, and Base Currency Withdrawal Amount are displayed when you click the Calculate button.
Base Currency Amount	Displays the total amount of the draft in the base currency.
Commission	Displays the commission applicable for the transaction.
Base Currency Withdrawal Amount	Displays the total amount of the draft in the base currency, plus commission.

- If required, waive the transaction fee. See, Section 5.3.4, "Waiving the Fee for a Transaction".
- Enter the appropriate Payment Method details. See, Section 5.3.3, "Entering Payment Method Details" section.
- 8. Click Submit.

# 5.16 Issuing a Cashier's Check

You use the Cashier Check operation to issue a check on behalf of a customer. The payment method is by cash, check, or funds transfer from the customer's account.

You can use preprinted stock, or you can issue a cashier's check dynamically. You enter the payment details and the fee payment details, or you can waive the fees as appropriate.

#### To issue a cashier's check

- 1. Navigate to the Customer Transactions > Cashiers Check screen (Fast Tran 213).
- Complete the Cashiers Check Details. The required fields are described in the following table.

Field	Comment
Issue Dynamically Allocated Serial Number	Select this radio button to automatically assign a serial number to the cashier's check.
Serial Number	Type the serial number of the check, if the cashiers check is being issued on pre-printed stock.
Payee Name	Type the name of the payee.
Amount	Type the amount of the check.
Date	Displays the date.
Memo	Type a memo, if required.

- **3.** If required, waive the transaction fee. See, Section 5.3.4, "Waiving the Fee for a Transaction".
- 4. Enter the appropriate Payment Method details. See, Section 5.3.3, "Entering Payment Method Details".
- **5.** Enter the authentication details for the customer, if required. See, Section 5.3.5, "Entering Customer Authentication Details".

**6.** Click Submit on the Issue Cashiers Check screen.

## 5.17 Issuing a Money Order

You use the Money Order operation to issue a money order on behalf of a customer. The payment method is by cash, check, or funds transfer from the customer's account.

You can use preprinted stock, or you can issue a money order dynamically. You enter the payment details and the fee payment details, or you can waive the fees as appropriate.

#### To issue a money order

- Navigate to the Customer Transactions > Money Order screen (Fast Tran 214).
- Complete the Money Order Details. The required fields are described in the following table.

Field	Comment
Issue Dynamically Allocated Serial Number	Select this radio button to automatically assign a serial number to the money order.
Serial Number	Type the serial number of the money order, if the money order is being issued on pre-printed stock.
Date	Displays the date.
Amount	Type the amount of the money order.
Memo	Type a memo, if required.

The Purchaser details contain the customer's details.

- Enter the sender details.
- Enter the payee details.
- If required, waive the transaction fee. See, Section 5.3.4, "Waiving the Fee for a Transaction".
- Enter the appropriate Payment Method details. See, Section 5.3.3, "Entering Payment Method Details".
- **7.** Click Submit on the Issue Money Orders screen.

### 5.18 Purchasing U.S. Savings Bonds

You use the Bond Purchase operation to purchase U.S. savings bonds (I or EE series) on behalf of a customer. The customer can pay for the purchase by cash, by check, or a combination of both. You can record the quantity and type of U.S. Bonds to be purchased, owner details (that is, ownership type, registration, tax ID number, ownership options, and additional names), and delivery and contact details.

**Note:** Bonds can be delivered to U.S. addresses only.

#### To purchase U.S. bonds

1. Navigate to the Customer Transactions > Bond Purchase screen (Fast Tran 215).

2. Complete the Bond Selection details. The fields are described in the following

Field	Comment
Series	Select I or EE.
Denomination	Select a denomination for the bond, for example, \$50.
Quantity	Type the number of bonds required.

**3.** Click Add to List.

The Bond List Table is updated with the bonds that you selected, and the Total Current Value and Total Number of Bonds fields are updated.

- 4. Enter the appropriate Payment Method details. See, Section 5.3.3, "Entering Payment Method Details".
- **5.** Click Next.
- Complete the Bond Owner Details. Some fields are described in the following table.

Field	Comment
Ownership Type	Select Individual or Fiduciary.
Full Name of Owner	Type the full name if you selected Individual as the ownership type.
Fiduciary Registration	Type the registration.
Social Security Number	Type the social security number if you selected Individual as the ownership type. If you selected Fiduciary, type the social security number or a taxpayer ID number.
Taxpayer ID Number	Type the social security number or a taxpayer ID number, if you selected Fiduciary as the ownership type.
Ownership Option	Select Co-Owner or Beneficiary or Single Owner.
Co-owner or Beneficiary Name	Type the full name of any co-owner or beneficiary of the bonds.

- **7.** Click Next.
- Complete the bond delivery details and contact information. By default, the fields contain the customer's details, but the bonds can be delivered to another person.
- Click Submit.

# 5.19 Making U.S. Savings Bond Inquiries and Redeeming Bonds

You use the Bond Inquiry operation to answer inquires about the redemption values of bonds for a customer, and to redeem bonds if required.

You can request a redemption value for the bonds up to a possible maximum of six months in the future. Future-dated redemption valuations depend on data from the U.S. Treasury being available for that date. The bond redemption values are downloaded from the U.S Treasury Web site using the Administration Console. You cannot redeem a bond within 12 months of its issue.

You require customer identification if the overall redemption value is more than the limit set by your bank, and if the customer has had an account with the bank for less than six months.

You make an inquiry for a maximum of 100 bonds for each transaction.

#### To make U.S. bond inquiries or redeem bonds

- Navigate to the Customer Transactions > Bond Redemption screen (Fast Tran 216).
- 2. Complete the Bond Selection details. The fields are described in the following table.

Field	Comment
Series	Select I or EE.
Denomination	Type a letter corresponding to the bond denomination. For example, type L for the \$50 denomination. The denomination letters vary depending on what has been configured for your bank.
Serial Number	Type a serial number if the customer wants to redeem bonds. The serial number is not required for inquiries
Issue Date	Type the issue date of the bond. This must be in the format: month/year, for example, 12/2002.

#### **3.** Click Add to List.

The Bond List Table is updated with the bonds that you specified.

Field	Comment
Serial Number	Displays the serial number of the bond.
Series	Displays the series; I or EE.
Denomination	Displays the bond denomination.
Issue Date	Displays the issue date of the bond.
Redemption Value	Displays the price at which the issuing company may choose to repurchase the bond before its maturity date.
Interest Earned	Displays the interest earned by the bond since the issue date.

- If you want to update details of the bonds, or redeem bonds, select the required bonds in the table.
- In the Value as of field, you can leave the current processing date, or type a date up to six months in the future. You must type this date in the correct format: month/year, for example, 08/2005.
- **6.** Click Calculate.
- The Redemption Value and Interest Earned are calculated and displayed.
- If the amount of the transaction exceeds the limit for the amount of bonds that can be redeemed, or if the customer has held an account for less than 6 months, click Authenticate Details, and enter the authentication details for the customer. (Details might already be provided from a previous transaction.)
- **9.** If required, click Redeem Bonds and check the details.
- **10.** Click Submit.

A receipt is printed with the redemption details at the end of the customer session.

# **Performing Customer Inquiries**

You perform the customer inquiries to obtain more information about the customer accounts. To perform a particular operation, you must have the necessary privilege.

This chapter covers the following topics:

- Section 6.1, "Displaying Account History"
- Section 6.2, "Maintaining Transfers"
- Section 6.3, "Maintaining Payments"
- Section 6.4, "Displaying Check Details and Stopping Checks"
- Section 6.5, "Displaying Credit Card Movements"
- Section 6.6, "Displaying the Currency Exchange Rates"
- Section 6.7, "Displaying the Interest Rates"

### 6.1 Displaying Account History

You use the Account History operation to deal with inquiries from a customer about movements on an account. Movements can be financial transactions such as deposits, withdrawals, direct debits, standing orders, payments, checks drawn, or charges.

You can specify a variety of search criteria to show details of the account for a period of time or for a previous number of transactions. The account details include the running balance on the account, which is calculated on a daily basis.

#### To search account history

- Navigate to the Customer Inquiries > Account History screen (Fast Tran 31).
- Enter the account details if they are not already displayed; see Section 5.3.1, "Selecting the Account".
- Use one of the following search criteria:
  - Select Date Range and enter the dates.
  - Select Specified Period, select a Period and type a number. For example, select Week and type 2 for all the transactions in the past two weeks.
  - Select Last and type a number of transactions to display the most recent transactions on the account.
- Click Account History.

The account history is displayed with the most recent transaction first. Note that the balance shown is the balance at the end of each day.

## 6.2 Maintaining Transfers

You use the Transfers operation to maintain the transfers from an account. You can specify a variety of search criteria to show payments for a period of time, or for a previous number of transfers. The payment details include the beneficiary name, branch code, account number, effective date, amount, status, and narrative.

You can display a list of pending transfers; that is, transfers to be made after today's date. You can delete or change the details of pending transfers.

#### To maintain transfers

- Navigate to the Customer Inquiries > Transfers screen (Fast Tran 32).
- Enter the account details if they are not already displayed; see Section 5.3.1, "Selecting the Account".
- Use one of the following search criteria to display the list of transfers:
  - Select Date Range and enter the dates.
  - Select Specified Period, select a Period and type a number. For example, select Week and type 2 for all the transfers in the past two weeks.
  - Select Last and type a number of transfers to display the most recent transfers from the account.
  - Select Pending Transfers to display all transfers that are to be made after today's date.
- **4.** Click Display Transfers.

The Transfer Details list is displayed.

- **5.** If you displayed a list of pending transfers, do one of the following as required:
  - To delete a pending transfer, select the transfer, click Delete, and click Submit.
  - To change the details of a pending transfer, select the transfer, click Amend, change the details of the transfer, and click Submit.

The Transfer Details list is updated.

Click Cancel when you are finished.

### 6.3 Maintaining Payments

You use the Payments operation to display the utility payments from an account.

You can specify a variety of search criteria to show payments for a period of time or for a previous number of payments. The payment details include the utility name, branch code, account number, date, amount, status, narrative, and reference number.

You can display a list of pending payments; that is, payments to be made after today's date. You can delete or change the details of pending payments.

#### To maintain payments

- Navigate to the Customer Inquiries > Payments screen (Fast Tran 33).
- Enter the account details if they are not already displayed; see Section 5.3.1, "Selecting the Account".
- Use one of the following search criteria to display a list of payments:
  - Select Date Range and enter the dates.

- Select Specified Period, select a Period and type a number. For example, select Week, and type 2 for all the payments in the past two weeks.
- Select Display Last, and type a number of payments to display the most recent payments from the account.
- Select Display Pending Payments to display all payments that are to be made after today's date.
- 4. Click Display Payments.

The Payment Details list is displayed.

- **5.** If you display a list of pending payments, do one of the following:
  - To delete a pending payment, select the payment, click Delete, and click Submit.
  - To change the details of a pending payment, select the payment, click Amend, change the details of the payment, and click Submit.

The Payment Details list is updated.

**6.** Click Cancel when you are finished.

# 6.4 Displaying Check Details and Stopping Checks

You use the Checks operation to deal with an inquiry from a customer about checks on an account, and to place a stop on a check.

You can search for checks by:

- Check Number
- Check Range
- Date Range
- Check Status, for example, Issued, In progress, Cleared, Stopped, Unknown, Awaiting dispatch, Cancelled, Printed, Received, Rejected and Returned.

You cannot stop a check if it has been stopped previously, or if it has passed through the clearing system.

### To display check details

- Navigate to the Customer Inquiries > Checks screen (Fast Tran 34).
- Enter the account details if they are not already displayed; see Section 5.3.1, "Selecting the Account".
- Enter the Search Criteria:
  - Click Check Number to display a specific check.
  - Click Check Range and enter dates in the From and To fields to display checks within the specified number range.
  - Click Check Date and enter numbers in the From and To fields to display checks within the specified date range.
  - Click Check Status, and select a status, to display all checks with that status.
- **4.** Click Display Checks.

A list of checks on the account is displayed.

- **5.** To stop a check, select the check in the list and click Stop. Select the reason for stopping the check.
- **6.** Click Cancel when you are finished.

### 6.5 Displaying Credit Card Movements

You use the Credit Card Transactions operation to display the history of a credit card account.

#### To display credit card movements

- Navigate to the Customer Inquiries > Credit Card Transactions screen (Fast Tran 35).
- Enter the account details if they are not already displayed; see Section 5.3.1, "Selecting the Account".
- 3. Select a credit card number.
- **4.** Use one of the following search criteria:
  - Select Date Range and enter the dates.
  - Select Specified Period, select a Period, and type a number. For example, select Week and type 2 for all the transactions in the past two weeks.
  - Select Last and type a number of transactions to display the most recent transactions on the account.
- **5.** Click Account History.

The account history is displayed with the most recent movement first.

## 6.6 Displaying the Currency Exchange Rates

Siebel Branch Teller has the ability to execute transactions in multiple currencies and provide Foreign Exchange related services to customers. The currency exchange rate is the rate at which one currency can be exchanged for another. You can view the prevailing currency exchange rate both in online and offline mode by navigating to Customer Inquiries > Exchange Rates (Fast Tran 36).

The fields are described in the following table.

Field	Comments
Currency	Displays the name of the currency.
We Buy Cash	Displays the rate at which the particular currency is bought.
We Sell Cash	Displays the rate at which the particular currency is sold.
We Buy Checks	Displays the rate at which Travelers Cheques are bought.
We Sell Checks	Displays the rate at which Travelers Cheques are sold.

### 6.7 Displaying the Interest Rates

You can view the prevailing interest rate of a product both in online and offline mode by navigating to Customer Inquiries > Interest Rates (Fast Tran 37). The ability to view interest rates is beneficial for the efficient and service optimizing interaction between you and the customer.

The fields are described in the following table.

Field	Comments
Product	Displays the name of the product.
Description	Displays the description of the product.
Interest Rate Type	Displays the interest rate type of the product. There can be multiple interest rate types for one product.
Rate	Displays the rate percentage of the product.

# **Performing Customer Services**

You use the Customer Services operations to maintain detail of customers, beneficiaries, and utility payments applying to their accounts.

This chapter covers the customer maintenance operations that you can perform. It includes the following topics:

- Section 7.1, "Maintaining Customer Details"
- Section 7.2, "Maintaining Beneficiary Details"
- Section 7.3, "Maintaining Utility Payment Details"
- Section 7.4, "Maintaining Standing Orders"
- Section 7.5, "Maintaining Direct Debits"
- Section 7.6, "Maintaining Overdrafts"
- Section 7.7, "Maintaining Statements"
- Section 7.8, "Maintaining Bank Cards"
- Section 7.9, "Issuing a Checkbook"
- Section 7.10, "Opening an Account"
- Section 7.11, "Closing an Account"
- Section 7.12, "Creating a Credit Card Application"

### 7.1 Maintaining Customer Details

You use the operation to set up, view, and maintain the details of customers. The details for a customer are divided into the following sections, indicated by tabs:

- **Personal.** The customer's name, date of birth, social security number, and so on
- **Dependents.** The number of dependents, their names, and date of birth
- **Employment.** The customer's employer and salary details
- **Residential.** The customer's current address, previous address, and contact details

You are required to enter only the personal details and some of the residential details for customers. The other details are optional, and you record the details only if the customer is agreeable to you doing so.

The customer must have an account before you can maintain customer details.

#### To maintain customer details

1. Navigate to the Customer Services > Customer Details screen (Fast Tran 41).

If you are in a customer session, the personal details for the customer are displayed, and you can change the details by clicking Amend. If you are not in a customer session, you can enter search details for an existing customer. If you are entering new personal details for a customer, complete the fields as described in the following steps.

- **2.** Enter or change the personal details for the customer. You must enter details in all fields except Middle Name. The Social Security Number and Mother's Maiden Name are required in the United States only.
- 3. Complete the Identification details. The fields are described in the following table.

Field	Comment
Authenticate Customer	Select the type of document used to verify the customer's identity.
Source of Address	Select the type of document used to verify the customer's address.
ID Number	Type the number of the document used to verify the customer's identity.
Address Reference	Type the reference number of the document used to verify the customer's address.

- 4. Click Next.
- **5.** Enter or change the details of dependents if required. To create the details of a dependent, click New, and enter the details as required. Click Next.
- Enter or change the customer's employment details, if required. You must select from Employment Status and Currency. Click Next.
- Enter or change the customer's residential details. You must select a Residential Status and Country, and enter details for at least Address Line 1, Zip/Postal Code, City, and Home Telephone Number. All other details are optional.
- 8. Click Submit.

### 7.2 Maintaining Beneficiary Details

You use the Beneficiaries operation to set up and maintain the details of beneficiaries for a customer account.

A beneficiary is a person or company to whom some form of payment is made from a customer's account. You use beneficiary account details in the setting up of standing orders, direct debit mandates, or funds transfers for a customer. When you set up the beneficiary details for a customer, they are used for all accounts held by that customer.

#### To maintain beneficiary details

- Navigate to the Customer Services > Beneficiaries screen (Fast Tran 42). The Beneficiary Details are displayed.
- **2.** Enter the customer's account details if they are not already displayed.
- Perform the required maintenance activity as follows:
  - To create a beneficiary, click New. Type the beneficiary name, branch code, account number, and, if required, a narrative. Finally, click Submit.

- To change the details of a beneficiary, select it from the list and click Amend. Change the details as required and click Submit.
- To delete a beneficiary, select it from the list, and click Delete. Click Submit.
- Click Close/Ok.

## 7.3 Maintaining Utility Payment Details

You use the Utilities operation to set up and maintain a list of utilities to which a customer can make payments.

Your bank keeps a list of utility companies and their banking details. For a customer to pay the utility bill, you must set up the utility for the customer. When you set up the utility details for a customer, they are used for all accounts held by that customer.

Before you perform the operation, an administrator must set up the list of utility companies and their details using the Administration Console.

#### To maintain utility details

- 1. Navigate to the Customer Services > Utilities screen (Fast Tran 43). The Payment Details are displayed.
- **2.** Enter the customer's account details if they are not already displayed.
- **3.** Perform the required maintenance activity as follows:
  - To create a utility, click New. Select a utility name. The branch code and account number are displayed. Type a reference, and, if required, a narrative. Finally, click Submit.

Field	Comment
Utility Name	Select the name of the utility.
Branch Code	The branch code that is displayed.
Account Number	The account number that is displayed.
Reference Number	The reference number of the utility.
Narrative	The description of the utility

- To change the details of a utility, select it from the list, and click Amend. Change the details as required, and click Submit.
- To delete the details of a utility, select it from the list, and click Delete. Click Submit.
- 4. Click Cancel.

# 7.4 Maintaining Standing Orders

You use the Standing Orders operation to set up and maintain standing orders on an account. In a standing order, the payment amount is fixed for each payment, unlike a direct debit where the payment amount can vary between payments. You can:

- Display all the standing orders on an account, whether they have a status of Active, Cancelled, or Suspended.
- Create a standing order, as long as this is permitted for the account.

- Change the details of a standing order.
- Cancel or delete a standing order.
- Suspend a standing order.
- Restart a suspended standing order.

### 7.4.1 To maintain standing orders

- 1. Navigate to the Customer Services > Standing Orders screen (Fast Tran 44).
- Enter the account details if they are not already displayed; see Section 5.3.1, "Selecting the Account".
- Perform the required maintenance activity as follows:
  - To create a standing order, click New. Type the Beneficiary details; that is, the details of the account to which payments will be made.

Field	Comment
Beneficiary Name	Select the name of the beneficiary of the standing order.
Branch Code	Displays the branch code.
Account Number	Displays the account number.
Beneficiary Narrative	Type a description.

Click Setup Beneficiary to add the beneficiary

Type the payment details, as described in the following table.

Field	Comment
Amount	Type the amount of the standing order
Frequency	Select the frequency for the standard order.
Total No. of Payments	Type the total number of payments.
No. of Payments to Date	Displays the number of payments made already.
First Payment Date	Displays the date of the first payment. This field displays the current processing date by default. Type a different date if required.
Next Regular Payment Date	Displays the date of the next payment.
Final Payment Date	Displays the date of the final payment.

#### Click Submit.

- To change a standing order, select it from the list and click Amend. Change the details as required and click Submit.
- To cancel a standing order, select it from the list and click Delete. Type the cancellation date. Select a cancellation reason. For example, Paid in Full, Customer Request, or Account Closed. Click Submit. The status of the standing order changes to Cancelled.
- To suspend a standing order, select it from the list and click Suspend. Type the suspension start date, the suspension end date and, select a suspension reason,

- for example, Unemployed. Click Submit. The status of the standing order changes to Suspended.
- To restart a suspended standing order, select it from the list and click Restart. Type the suspension end date and click Submit. The status of the standing order changes to Active.
- **4.** Click Cancel when you are finished.

# 7.5 Maintaining Direct Debits

You use the Direct Debit operation to set up and maintain direct debit mandates on an account. You can:

- Display all the direct debit mandates on an account, whether they have a status of Active, Cancelled, or Suspended.
- Create a direct debit mandate, as long as this is permitted for the account.
- Change the details of a direct debit mandate.
- Cancel a direct debit mandate.
- Suspend a direct debit mandate.
- Restart a suspended direct debit mandate.

#### To maintain direct debits

- Navigate to the Customer Services > Direct Debits screen (Fast Tran 45).
- Enter the account details if they are not already displayed; see Section 5.3.1, "Selecting the Account".
- **3.** Perform the required maintenance activity as follows:
  - To create a direct debit mandate, click New. Type the Originator details; that is, the details of the account to which payments will be made.

Field	Comment
Branch Code	Displays the branch code.
Account Number	Displays the account number.
Originator Name	Type the name for the account to which the
Reference Number	Type a reference name for the direct debit.

Type the direct debit details, as described in the following table.

Field	Comment
Frequency	Select the frequency for the direct debit.
First Payment Date	Displays the date of the first payment. This field displays the current processing date by default. Type a different date if required.
Final Payment Date	Displays the date of the final payment.

Finally, click Submit.

To change a direct debit mandate, select it from the list and click Amend. Change the details as required and click Submit.

To cancel a direct debit mandate, select it from the list, and click Delete. Type the cancellation date. If required, select a cancellation reason. For example, Paid in Full, Customer Request, or Account Closed. Click Submit.

The status of the direct debit mandate changes to Cancelled.

To suspend a direct debit, select it from the list and click Suspend. Type the suspension start date and, if required, select a suspension reason, for example, Unemployed. Click Submit.

The status of the direct debit mandate changes to Suspended.

To restart a suspended direct debit, select it from the list and click Restart. Type the suspension end date and click Submit.

The status of the direct debit mandate changes to Active.

**4.** Click Cancel when you are finished.

### 7.6 Maintaining Overdrafts

You use the Maintain Overdraft operation to set up and maintain the details of the overdraft on a customer's account. You can do the following:

- Display the details of the overdraft facility.
- Create a new overdraft facility on an account, as long as this is permitted for the account.
- Change the details of an overdraft facility.
- Cancel an overdraft facility.

#### To maintain overdraft details

- Navigate to the Customer Services > Overdrafts screen (Fast Tran 46).
- **2.** Enter the account details if they are not already displayed; see Section 5.3.1, "Selecting the Account".
- **3.** Perform the required maintenance activity as follows:

Field	Comment
Amount	Type the amount for which the overdraft is created.
Start Date	Select the date on which the overdraft is applicable.
End Date	Select the date till which the overdraft is applicable. End date should be greater than the start date.
Reason	Select the mandatory reason for overdraft.

- To create an overdraft, type the amount, the start date and end date and select a reason. Click Submit.
- To change the overdraft, click Amend. Change the details as required, and click Submit.
- To cancel the overdraft, click Cancel Overdraft. Type the cancellation date, if this is different from the displayed processing date, and, if required, select a reason for cancellation, for example, Customer Request, Account Closed, or Paid in Full. Click Submit.

When you click Cancel, the operations home page is displayed.

### 7.7 Maintaining Statements

You use the Statements operation to maintain the details of statements for an account. You can change the statement frequency, and also stop the dispatch of a statement if the customer requests not to receive statements.

You can specify that statements are sent by the postal service, by fax, or by email. In most cases, the statement is sent to the customer's home address, fax number, or email address. However, the customer can request that you send it to another address, for example, the statement address. A copy of the statement can also be sent to another address (perhaps, an accountant).

In the case of joint accounts, each holder can receive a statement.

#### To maintain statements

- Navigate to the Customer Services > Statements screen (Fast Tran 47).
- Enter the account details if they are not already displayed; see Section 5.3.1, "Selecting the Account".
- Complete the Statement Details. The fields are described in the following table.

Field	Comment
Statement Frequency	Select the frequency with which statements are to be sent.
Statement Per Customer	Select this check box if the account is a joint account, and a statement is to be sent to each account holder.
Send Copy to Different Address	Select this check box if a copy of the statement is to be sent to another person or address.
Last Statement Date	The date on which the last statement was sent.
Next Statement Date	The date on which the next statement will be sent.
Stop Statement	Select this check box to stop sending statements for this account.

#### **4.** Select a method of correspondence:

- Click Mail if the statement is to be posted. If the statement is to be sent to a different address from the customer's address, click Amend and type the required details in the Account Statement Details. Select, **Use the Selected** Customer Details as the Account Statement Details, if you want to send the account statements to the existing mail address given by the customer, in the bank. Note that, you will see this checkbox only if you select Mail in Method of Correspondence. Sometimes, you need to scroll down to see this checkbox.
- Click Fax, and type a fax number if the statement is to be sent by fax.
- Click Email Address and type an address if the statement is to be sent by
- **5.** If you selected Send Copy to Different Address, type the address details in Copy of Statement Details.
- Click Submit.

## 7.8 Maintaining Bank Cards

You use the Bank Cards operation to maintain the details of bank cards used with an account, and to request the issue of new bank cards.

Your bank can issue a number of different bank cards for an account, for example:

- ATM card
- Debit card
- Check guarantee card

Only certain account types allow the issue of a card.

#### You can:

- Request the issue of a new bank card for the account. Siebel Branch Teller obtains and allocates a card number, generates a PIN, and calculates the expiry date for the new card.
- Cancel a bank card when it has been lost, stolen, damaged, withdrawn or returned.
- Reissue a bank card that is near to its expiry date. Siebel Branch Teller generates a new PIN and calculates the next expiry date.
- Change the maximum cash withdrawal amount for the card.
- Reset the number of tries on the PIN card counter to zero. You do this for a customer who has exceeded the number of PIN tries allowed for the customer.
- Reissue a PIN for an existing bank card.

#### To maintain bank cards

- Navigate to the Customer Services > Bank Cards screen (Fast Tran 48).
- Enter the account details if they are not already displayed; see Section 5.3.1, "Selecting the Account".
- In the Card Details list, perform the required action:
  - To issue a new card, click New, select the card type and card holder, and click Submit.

Field	Comment
Card Type	Select the type of card, for example, ATM card, Debit card.
Card Holder	Select a name from the list.

- To reissue a PIN, select a card from the list, and click Re-Issue PIN.
- To reset a PIN counter, select a card from the list, and click Reset PIN Tries.
- To reissue a card, select a card from the list, and click Re-Issue Card.
- To change a withdrawal limit, select a card from the list, and click Amend Limit. Type the required amount in Cash Withdrawal Limit and click Submit.
- To cancel a card, select a card from the list, and click Cancel Card. Click one of the following: Lost, Stolen, Damaged, Withdrawn, or Returned. Type the cancellation details, and click Submit.
  - Depending on the action, the Card Details list is updated.
- **4.** Click Cancel when you have finished.

## 7.9 Issuing a Checkbook

You use the Check Book operation to order a checkbook for a customer. If it is permitted to set up a checkbook facility for the account, you record the checkbook type details, and set up its method of dispatch.

The methods of dispatch for a checkbook are either by post to the customer's address, or for picking up in any of the branches. If the account is a joint account, the dispatch address is the address of the first account holder.

#### To issue a checkbook

- Navigate to the Customer Services > Check Book screen (Fast Tran 49).
- Enter the account details if they are not already displayed; see Section 5.3.1, "Selecting the Account".
- Enter the Check Details. Some fields are described in the following table.

Field	Comment
Account Name	Displays the customer's name.
Checkbook Type	Select the type of checkbook required.
Dispatch Method	Select Post or Collect at Branch.
Address	Type the address if you selected Post as the method of dispatch.
Collection at Branch Code	Type the branch code at which the checkbook is to be collected.

To change the check details of except the account name, click Amend.

Click Submit when you are finished.

## 7.10 Opening an Account

You use the Open Account operation to open a single or joint account for a customer. If the person who wants to open the account is not an existing customer, you must set them up as a customer on Siebel Branch Teller before opening the account process.

A product comparison screen is displayed, which allows you to discuss with the customer the types of products your bank offers. The information displayed about each product includes:

- Product Name
- Card Facility
- Minimum Balance
- Checkbook Facility
- Overdraft Facility
- **Interest Rate**
- Tax Rate

When you select the required account, Siebel Branch Teller allocates an account number and branch code to the customer.

A list of ancillary sales that are permitted to be made on the type of account is displayed. This allows you to sell additional services for the new account, and set up direct debits and payments from the account. The appropriate screens for you to set up the ancillary sales are displayed.

#### To open an account

- Navigate to the Customer Services > Open Account screen (Fast Tran 410).
- Select Single Account or Joint Account.
- Click Submit.
- The Customer Details: Personal Details screen is displayed. Enter or change the personal details for the customer. You must enter details in all fields apart from Middle Name.
- **5.** Complete the Identification details. The fields are described in the following table.

Field	Comment
Authenticate Customer	Select the type of document used to verify the customer's identity.
Source of Address	Select the type of document used to verify the customer's address.
ID Number	Type the number of the document used to verify the customer's identity.
Address Reference	Type the reference number of the document used to verify the customer's address.

- **6.** Click Next.
- Enter or change the details of dependents if required. To create the details of a dependent, click New, and enter the details as required. Click Next.
- Enter or change the customer's employment details, if required. You must select from Employment Status and Currency. Click Next.
- Enter or change the customer's residential details. You must select a Residential Status and Country, and enter details for at least Address Line 1, Address Line 2, Zip/Postal Code, City, and Home Telephone Number. All other details are optional.
- 10. Click Submit

The Open Account: Product Type screen is displayed.

11. Complete the Product Details. The fields are described in the following table.

Field	Comment
Product Type	Select the type of product, for example, Check, Savings.
Account Type	Select the type of account, for example, Personal, Small Business.
Account Currency	Select the currency for the account.

**12.** Complete the Mandate Instructions. The fields are described in the following table.

Field	Comment
Customer Name(s)	Displays the names of the customer or customers, in the case of a joint account
All to Sign	Click if all of the holders of the account should sign

Field	Comment
Either to Sign	Click if either of the holders of the account can sign
Account Name	Type the name of the account.

#### 13. Click Submit.

The New Account Details screen is displayed. The branch code, account number, account name, and product name for the new name are displayed. An ancillary sales list is also displayed.

- 14. Select the checkbox corresponding to the type of ancillary sale required and enter the required details:
  - To set up a bank card, click Bank Cards. For more information, see Section 7.8, "Maintaining Bank Cards".
  - To set up a checkbook, click Checkbook. For more information, see Section 7.9, "Issuing a Checkbook".
  - To set up an overdraft, click Overdraft. For more information, see Section 7.6, "Maintaining Overdrafts".
  - To set up a direct debit, click Direct Debits. For more information, see Section 7.5, "Maintaining Direct Debits".
  - To set up a utility payment, click Utility Payment. For more information, see Section 6.3, "Maintaining Payments".
  - To set up a standing order, click Standing Orders. For more information, see Section 7.7, "Maintaining Statements".
  - To change the default details for the statements for the account, click Statements.

#### **15.** Click Complete.

### 7.11 Closing an Account

You use the Close Account operation to close a customer's account. Before you try to close an account, check that the account can be closed, and check the account status. You cannot close accounts with a status other than active or dormant immediately. For example, frozen accounts require further investigation before you can close them.

When you close an account you can check:

- That no un-cleared funds are outstanding on the account.
- The current balance of the account. All deposit accounts should have a positive balance whereas current accounts can have a positive or negative balance. If the account balance is negative, you can transfer funds from another of the customer's accounts to clear the balance on the account being closed.
- Whether there are any outstanding charges due on the account. You can trigger the posting of these charges.
- Whether there is any outstanding interest on the account.
- Whether there are active ancillary services, such as bank cards, standing orders, or automated direct debits on the account being closed. Advise the customer that these ancillary services are closed along with the account.

When you close an account you can do one of the following with the balance of

- Transfer the balance to another account held by the customer in your bank
- Transfer the balance to an account held in same bank
- Pay the balance to the customer as cash You can request a closing statement for the account.

#### To close an account

- Navigate to the Customer Services > Close Account screen (Fast Tran 411).
- 2. Enter the account details if they are not already displayed; see Section 5.3.1, "Selecting the Account".

All ancillary services associated with the account are listed. All ancillary services are cancelled when the account is closed.

- Click View to display details of a particular ancillary service. Click Continue.
- **4.** Complete the Account Details. Some fields are described in the following table.

Field	Comment
Opening Balance	Displays the opening balance of the account.
Interest Accrued	Displays any interest accrued on the account. Click Capitalize Interest to add the interest to the balance.
Outstanding Charges	Displays the outstanding charges on the account. Click Apply Charges to apply the charges to the account.
Closing Balance	Displays the closing balance of the account, which is calculated from the opening balance, interest, and charges applied.

- 5. If you want to waive any of the charges on the account, click Waive Charges, and type the amount you want to waive.
- 6. Select a Method of Settlement, for example, Funds Transfer, Withdrawal, Inter Account Transfer.
- 7. Click Settlement.

Depending on the method of settlement that you selected, a screen is displayed that allows you to make, for example, the necessary funds transfer, withdrawal, or inter-account transfer. After you have completed the appropriate financial transaction, the Account Details are updated.

- 8. Select a Closure Reason, for example, Account Holder Relocation, Death of Account Holder.
- 9. Click Submit.

The account is removed from the customer's list of accounts.

### 7.12 Creating a Credit Card Application

You use the Credit Card operation to create a credit card application for a customer, and to display details of existing credit cards for a customer.

You can search for a customer's credit card details by specifying an account number, or a credit card number and expiry date.

You can create a credit card application for an existing customer only. The application contains the following sections:

- **Personal.** The customer's name, date of birth, social security number, and so on
- **Dependents.** The number of dependents, their names, and dates of birth
- **Additional Card Holder.** The first name and last name
- **Residential.** The customer's current address, previous address, and contact details
- **Employment.** The customer's employer and salary details
- **Financial.** Card type, salary details, monthly expenses, and bank details To create a credit card application
  - 1. Navigate to the Customer Services > Credit Card Application screen (Fast Tran 412).
  - Enter the account details if they are not already displayed; see Section 5.3.1, "Selecting the Account".
  - 3. Enter or change the personal details for the customer. You must enter details in all fields apart from Middle Name. The Social Security Number and Mother's Maiden Name are required in the United States only.
  - 4. Complete the Identification details. The fields are described in the following table.

Field	Comment
Authenticate Customer	Select the type of document used to verify the customer's identity.
Source of Address	Select the type of document used to verify the customer's address.
ID Number	Type the number of the document used to verify the customer's identity.
Address Reference	Type the reference number of the document used to verify the customer's address.

- **5.** Click Next.
- **6.** Enter or change the details of dependents, if required. To create the details of a dependent, click New, and enter the details as required. Click Next.
- 7. Enter or change the details of additional cardholders, if required. Click Next.
- Enter or change the customer's residential details. You must type the date that the customer moved in, enter details for at least Address Line 1 and Home Telephone Number. All other details are optional. Click Next.
- **9.** Enter or change the customer's employment details, if required. You must select from Employment Status, and type a job title, an employment start date, and telephone number. Click Next.
- 10. Complete the customer's financial details. Some fields are described in the following table.

Field	Comment
Card Type	Click the radio button corresponding to the type of card being applied for.

Field	Comment
Salary Details	Type or select the salary details as required. You must select a currency.
Outstanding Loans	Type the monthly amount that the customer pays for outstanding loans.
Time With Bank	Type the number of years and months that the customer has been with the bank.
Disclose Information	Select this check box if the customer wants to have his or her personal information disclosed to other parties.
Card Protection Plan	Select this check box if the customer wants to avail of a card protection plan, and click Direct Debit to set up a direct debit for the plan.

#### **11.** Click Submit.

# **Performing Teller Operations**

You use the Teller operations to perform operations that do not involve customer accounts. You cannot perform the operations in a customer session.

This chapter covers the teller operations that you can perform. It includes the following topics:

- Section 8.1, "Starting Your Day"
- Section 8.2, "Performing Interim Balance Inquiries"
- Section 8.3, "Performing Batch Inquiries"
- Section 8.4, "Opening Cashboxes/Drawers during Signon"
- Section 8.5, "The Journal Activities"
- Section 8.6, "Reconciling Failed Transactions"
- Section 8.7, "Debiting or Crediting a General Ledger Account"
- Section 8.8, "Transferring Cash Between Cashboxes"
- Section 8.9, "Transferring Checks Between Cashboxes"
- Section 8.10, "Transferring Ownership of a Cashbox"
- Section 8.11, "Closing Open Batches of Checks"
- Section 8.12, "Transferring a Checkbook"
- Section 8.13, "Ending the Business Day"
- Section 8.14, "Performing End-of-Day Processing"
- Section 8.15, "Logging Out"

### 8.1 Starting Your Day

Before you can process financial transactions, Siebel Branch Teller must assign a cashbox to you If you are a primary holder of the cashbox, you must record the totals for each currency in your cashbox. When you log in to Siebel Branch Teller after someone has performed the Branch Start of Day operation, you must start your day by selecting a cashbox assignment.

Siebel Branch Teller can assign you as non-primary holder of a cashbox if there is already a primary holder of the cashbox. As a non-primary holder, you are not responsible for cash balancing of the cashbox.

If the branch is operating in AM processing mode, Siebel Branch Teller creates a journal for you, which records all the transactions you perform for the current

processing date. If the branch is operating in AM/PM processing mode, the journal might already exist. Siebel Branch Teller prints a record of the starting positions; that is, the cash totals at the start of the day.

### 8.1.1 To start your day

- **1.** Navigate to the Teller Operations > Teller Start of Day screen (Fast Tran 51).
- **2.** Select a cashbox and holder position as described in the following table.

Field	Comment	
Previously Assigned Cashbox	Select this option to be assigned your most recently used cashbox.	
Assign New Cashbox Single Mode	Select this option to be assigned a currently unassigned cashbox and to be the primary holder of that cashbox.	
Assign New Cashbox Shared Mode	Select this option to operate a cashbox in shared mode. You can choose one of the following:	
	<ul> <li>To be assigned a currently unassigned cashbox, and to be the primary holder of that cashbox</li> </ul>	
	<ul> <li>To be assigned a currently assigned cashbox, and to be the non-primary holder of that cashbox</li> </ul>	
Primary Holder	Select this option to be assigned as the primary holder of the cashbox. You can operate in single mode or shared mode.	
Non-Primary Holder	Select this option to be assigned as the non-primary holder of a currently assigned cashbox. The primary holder must have already been assigned and you must operate in shared mode.	
Disable Cashbox Transactions	Select this option if you need to perform non cash transactions. Following sub menus under Transactions menu are disabled for a cashless user.	
	<ul><li>Deposit</li></ul>	
	<ul><li>Withdrawal</li></ul>	
	<ul><li>Cash Check</li></ul>	
	<ul> <li>Issue Draft</li> </ul>	
	<ul> <li>Issue Cashiers' Check</li> </ul>	
	<ul> <li>Issue Money Order</li> </ul>	
	■ FX Issue Draft	
	<ul><li>Cash Exchange</li></ul>	
	<ul> <li>Foreign Exchange</li> </ul>	
	In the Payments submenu, the option to fund the payment by cash or check is disabled.	

Field	Comment	
Enable Cash Transactions	Select this option if you want to perform cash related transactions after being a cashless user. You cannot end the customer session. You need to transfer the incomplete customer session to a teller with cashbox to complete it. When the session is transferred to target teller with cashbox, notification is prompted.	
	You can make only cash deposit and not check deposit, if you select this option.	
	Following sub menus under Customer Transactions menu are disabled.	
	<ul><li>Cash Check</li></ul>	
	<ul><li>Local Draft</li></ul>	
	<ul><li>Cashiers' Check</li></ul>	
	<ul><li>Money Order</li></ul>	
	<ul> <li>Foreign Issue Draft</li> </ul>	
	■ Cash Exchange	

- Click Submit. Depending on the radio buttons you selected, the Assign Cashbox screen, the Cash Details screen, or operations home page is displayed as follows:
  - If you select Previously Assigned Cashbox and Primary Holder, you are assigned your most recently used cashbox, and the Cash Details screen is displayed.
  - If you select Previously Assigned Cashbox and Non-Primary Holder, you are assigned your most recently used cashbox that has already been assigned a primary holder. The operations home page is displayed, because you are not responsible for recording or reconciling the cashbox cash totals. The operation is now complete.
  - If you select Assign New Cashbox Single Mode and Primary Holder, the Assign New Cashbox Single Mode screen is displayed with a list of the cashboxes available for Single Mode operation. Select the cashbox you want to be assigned, and click Submit to complete the cashbox assignment. The Cash Details screen for the assigned cashbox is then displayed.
  - If you select Cashless User ((Disable Cashbox Transactions), your Journal is created or updated. You will be transitioned to the home page.
  - If you select Cashless User (Enable Cashbox Transactions), your Journal is created or updated. You will be transitioned to the home page.
  - If you select Assign New Cashbox Shared Mode and Primary Holder, the Assign New Cashbox Shared Mode/Primary Holder screen is displayed with a list of the cashboxes available for Shared Mode operation. Select the cashbox you want to be assigned, and click Submit to complete the cashbox assignment. The Cash Details screen for the assigned cashbox is then displayed.
  - If you select Assign New Cashbox Shared Mode and Non-Primary Holder, the Assign New Cashbox Shared Mode/Non-Primary Holder screen is displayed with a list of the cashboxes available for operating in shared mode (the primary holder has already been assigned). Select the cashbox you want to assigned, and Click Submit to complete the cashbox assignment. The non-primary holder is not responsible for recording or reconciling the cashbox

cash totals. The operations home page is displayed, and the operation is

- 4. If the Cash Details screen is displayed, you must now enter the cash totals for each currency in the cashbox.
- 5. Type the number of notes or coins for each denomination in the Number column, or the total in the Total column. If you enter the Number value, the corresponding Total field is automatically updated, or if you enter the Total value, the corresponding Number field is automatically updated.
- **6.** Click Confirm Totals to confirm if there is any difference between the system-generated totals and the actual totals.
  - Siebel Branch Teller calculates and displays the values for Total Notes, Total Coins, Total Cash, System Cash, and Difference.
- Click Next to enter the denomination details for the next currency.
- Repeat steps 5 through 7 for each currency in your cashbox.
- Click Submit to save the cashbox denomination details.

You have now completed the Teller Start of Day operation, and you are returned to the operations home page for transaction processing.

### 8.1.2 Available Teller Transactions

This section contains a table that shows the transactions that are possible for a teller with cashbox, cashless teller, and cashless teller with cash transactions.

	Session		Teller		
	Transactional	Customer Session	With Cashbox	Cashless	Cashless with Cash Transactions
Deposit	Yes	Yes	Yes		Yes
Withdrawal	Yes	Yes	Yes		Yes
Cash Check	Yes	Yes	Yes		
Inter-Account Transfer		Yes	Yes	Yes	Yes
Funds Transfer	Yes	Yes	Yes	Yes	Yes
Wire Transfer	Yes	Yes	Yes		
International Transfer	Yes	Yes	Yes		
Bill Payment	Yes	Yes	Yes	Yes (no cash or checks)	Yes (no checks)
Cash Exchange	Yes	Yes	Yes		
Foreign Exchange	Yes	Yes	Yes		Yes
Local Draft	Yes	Yes	Yes		
Foreign Draft	Yes	Yes	Yes		

	Session		Teller		
	Transactional	Customer Session	With Cashbox	Cashless	Cashless with Cash Transactions
Cashier's Check	Yes	Yes	Yes		
Money Order	Yes	Yes	Yes		
Bond Purchase	Yes	Yes	Yes		
Bond Redemption	Yes	Yes	Yes		
General Ledger	Yes	Yes	Yes	Yes	Yes

### 8.2 Performing Interim Balance Inquiries

You use the Interim Balance Inquiry operation to check the balance of your cashbox against the system-generated balance for the cashbox.

Siebel Branch Teller calculates the difference, if any, between your cashbox cash total and the system-calculated cashbox cash total. Siebel Branch Teller creates a journal entry of both the system-calculated and the actual denomination values.

#### To perform an interim balance inquiry

- Navigate to the Teller Operations > Interim Balance screen (Fast Tran 52).
- Type the quantity of notes or coins for each denomination in the Number column, or the total in the Total column. If you enter the Number value, the corresponding Total field is automatically updated, or if you enter the Total value, the corresponding Number field is automatically updated.
- **3.** Click Confirm Totals to confirm if there is any difference between the system-generated totals and the actual totals.
  - Siebel Branch Teller calculates and displays the values for Total Notes, Total Coins, Total Cash, System Cash, and Difference.
- **4.** Click Next to enter the denomination details for the next currency.
- **5.** Repeat steps 2 through 4 for each currency in your cashbox.
- **6.** If there is a difference between the amount you enter and the system-generated amount, do one of the following:
  - Enter the correct denominations to balance.
  - Enter the account number and branch code of an internal account. A balancing transaction is performed against this account.
- 7. Click Submit to save the cashbox denomination details.

## 8.3 Performing Batch Inquiries

You use the Batch History operation to retrieve negotiable instrument batch details. You can search for batches by batch ID or batch amount. If you are a supervisor, you can search for batches for any user.

The batch details retrieved include branch code, cash drawer ID, batch ID, batch status, number of items, check type, and batch currency. You can also display details of individual, negotiable instrument items in the batch. The details displayed include date, destination branch code, account number, check number, and check amount.

Before you start, you must know the batch ID or batch amount.

#### To perform a batch inquiry

- 1 Navigate to the Teller Operations > Batch History screen (Fast Tran 53).
- 2 Type a date in the Batch Date field, if you want to search for a date other than the current processing date.
- 3 Select the Teller Id from the drop-down list. If you are a teller, you can select only your own ID. If you are a supervisor or branch manager you can select any teller ID.
- 4 Click Batch Id or Batch Amount, and type the batch ID or batch amount respectively.
- **5** Click Display Batch Details.
- 6 If you want to display details for a batch item, select the item, and click Batch Item Details.
- 7 Click Previous to return to the Query Check Batch screen.
- 8 Click Cancel on the Query Check Batch screen to return to the operations home page.

### 8.4 Opening Cashboxes/Drawers during Signon

Siebel Branch Teller provides software cashboxes that represent the physical cashboxes that you use. There are two types of cashbox:

- **Standard cashbox.** Represents the cashbox that you use for your normal operations.
- Branch vault cashbox. Represents your branch vault, and is assigned only to users with appropriate privileges.

Each cashbox contains a currency box for each of the currencies that your bank uses. Each currency box can contain denominations (notes and coins) of a single currency. You are allowed to perform cash transactions in any currency for which you have a currency box. A cashbox can also contain negotiable instrument boxes for each of the negotiable instruments in use at your bank (for example, checks, traveler's checks, U.S. savings bonds, and money orders).

You can share a cashbox with another teller. In this case, you can either be the primary owner or the non-primary owner. The primary owner is responsible for recording or reconciling the cash totals in the cashbox; the non-primary holder is not responsible for the cash totals. If you are the primary owner of a cashbox, you can transfer the ownership of the cashbox to the non-primary owner, who then becomes the primary owner.

### 8.5 The Journal Activities

For each processing day, Siebel Branch Teller creates an electronic journal for you that record the following activities:

- Start of day
- All financial transactions
- Interim balancing

- Cashbox transfers
- End of day

Depending on your user role, you might be able to view your journal only, or in addition, the journals for other users in your branch. For example, if you are a supervisor, you can view details for the tellers that you supervise.

You can reverse financial transactions from the journal. You use the Journaling operation to view the journal, see Section 8.5.1, "Displaying the Journal, Reversing Transactions, and Printing a session receipt".

You can select an entry in the journal summary (see the following table) and display the details for that entry.

Journal Summary Fields

Field	Comment
Date/Time	Displays the date and time at which the transaction was posted.
Туре	Displays the type of the transaction.
Status	Displays the status of the transaction.
Teller	Displays the name of different tellers under the supervisor.
Amount	Displays the amount of the transaction.
Cashbox Id	Displays the id of the cashbox.

The fields in the Journal Details screen are shown in the following table. There can be several entry items for a transaction.

**Journal Details Fields** 

Field	Comment
JOURNAL ENTRY DETAILS	
Transaction Type	Displays the type of the transaction.
Currency	Displays the currency of the transaction.
Amount	Displays the amount of the transaction.
JOURNAL ENTRY ITEM DETAILS	
Payment Method	Displays the means of payment, for example, cash or check. Funds Transfer is another payment method. In journal entry, it is shown as Transfer.
Amount	Displays the amount of the entry item.
Currency	Displays the currency of the entry item.
Status	Displays the status of the check used for payment, for example, cleared or un-cleared.
Source Account	Displays the account number from which the payment was made.
Destination Account	Displays the account number to which a payment or transfer is made.
Negotiable Instrument No	Displays the number of a check used for payment.
STATUS	

Field	Comment
Status	Displays the status of the transaction, for example, ACCEPTED.
Online/Offline	Displays whether the transaction was performed in the online or offline processing mode.
Posting Date/Time	Displays the data and time at which the transaction was posted.
Override	Displays whether a supervisor override was requested for the transaction, together with the ID of the supervisor.
Fee Amount	Displays the amount of any fee associated with the transaction, and whether the fee was waived.

### 8.5.1 Displaying the Journal, Reversing Transactions, and Printing a session receipt

You use the Journaling operation to search for a specific journal entry, view the details of a journal entry, reverse a transaction in the journal, and print a session receipt.

The journal records the following types of operations for each user for each processing day: financial transactions, cashbox cash transfers, and reversal operations.

If you are a teller, you can retrieve and view your own journal entries based on a specified date range and one of the following search criteria: transaction type or transaction amount. If you are a supervisor, you can view the journal entries for tellers that you supervise.

You can reverse a transaction in the journal provided that:

- The user ID of the transaction in the journal is the same as your user ID.
- Your current processing date is the same as the journal date.
- The transaction status is Cleared or Pending.
- It is permitted to reverse the type of transaction.

When you successfully reverse a transaction, Siebel Branch Teller sets the status of the original journal entry and the status of the original transaction to Reversed. If cash was tendered in the original transaction, Siebel Branch Teller debits or credits the cashbox with the reversal amount.

**Note:** You can also reverse transactions through the transaction register in a customer session; see Section 5.1, "Customer Sessions".

You can search for currency transaction (CTR) reports in the journal. To do so, you search for a transaction type of CTR Suspended or CTR Complete. When you display the details for suspended CTRs, the CTR Processing screen is displayed, and you can resume the processing of the CTR form. For more information, see Section 5.2, "Currency Transaction Reporting".

Siebel Branch Teller collects the transactions performed for a customer during their interaction with the teller. When the session is ended, a receipt is automatically printed detailing the transactions performed. Sometimes, due to printer failure, the receipt may not be printed correctly or a user may require a copy for administrative reasons. It is possible to reprint the session receipt.

To display the journal and reverse transactions

- Navigate to the Teller Operations > Journal (Fast Tran 54).
- Complete the search criteria. The fields are described in the following table.

Field	Comment	
Teller	Select the teller, whose journal you are authorized to access.	
Transaction Type	Select a transaction type if you want to use this search criteria. Searching can be performed across All Transaction Types, All Customer Transactions, All Teller Transactions, and so on.	
Date Range	Type the From and To range of the date. By default both the fields contain the current processing date.	
Transaction Amount	Type the From and To range of the transaction amount.	
Currency	Select the currency.	
Narrative	Type a description, if required.	
Exact Match Only	Select this if you want to get journal search results only for the exact match that you have specified.	
Cashbox Id	Select the wanted cashbox id.	
Supervisor Override	Select this if you want to get journal search results based on supervisor override.	
Fees	Select this if you want to get journal search results based on fees.	
Negotiable Instruments	Select this if you want to get journal search results based on negotiable instruments.	

Click Search when you have entered the required search criteria.

The Journal Details are displayed, need to select Details button for more detailed information on the transaction.

- You can view the details of an entry in more detail and select Reverse button. If you want to reverse a transaction in the journal, select the appropriate entry and click Reverse:
  - If the transaction involved cash, the cash details screen is displayed. Enter the denomination details of the cash to be returned to the customer.
  - Click Submit to complete the reversal operation.

If a negotiable instrument was tendered in the original transaction, the status of the negotiable instrument is set to Returned, if the status of the original negotiable instrument was Received. Otherwise the status is set to Canceled, if the original status was Issued.

- If you want to view the details of a journal entry, click Details.
  - The Journal Entry Details screen is displayed. You can click Reverse on this screen to reverse the transaction.
- Click Previous to return to the Journal Summary screen.
- Click Cancel on the Journal Summary screen to return to the operations home page.

#### To reprint a session receipt

Navigate to the Teller Operations > Journal screen (Fast Tran 55).

- 2. Select the Transaction Type as End Of Customer Session. Complete the search
- **3.** Click Search when you have entered the required search criteria.
  - The Journal Details are displayed.
- 4. Select the journal entry for which you want to have a receipt and Click Details. The Journal Entry Details screen is displayed. You can click Print on this screen to print the session receipt.
- **5.** Click Previous to return to the Journal Summary screen.
- **6.** Click Cancel on the Journal Summary screen to return to the operations home page.

### 8.6 Reconciling Failed Transactions

You use the Reconciliation operation to reconcile failed transactions after a period of offline processing. Some transactions that have been processed while in offline processing mode can fail when forwarded to the central server. Such transactions might fail because of, for example, invalid account numbers, inactive account, closed account or insufficient funds available in account.

You can reconcile failed transactions at any time but, if you attempt a Teller End of Day operation, or an Interim Balance Enquiry operation, and there are failed offline transactions, you are prompted to reconcile them. A new journal entry is created for the reconciliation.

A supervisor can reconcile a teller's failed transactions. This can happen where the offline period has spanned more than one day and the teller who created the offline requests is not available.

#### To reconcile failed transactions

- 1. Navigate to the Teller Operations > Reconciliation screen (Fast Tran 55).
- **2.** Select a failed transaction.
- Click Details if you want to display reconciliation details.
- Click Reconcile. Type some comments if required and type an internal account number that the deposit is to be credited to, or debited from, to balance the transaction.
- Click Close.

### 8.7 Debiting or Crediting a General Ledger Account

You use the General Ledger operation to create a debit or credit transactions between source and destination accounts. It is possible to perform a back dated and future dated transaction. This transaction enables you to do one of the following.

- Credit a general ledger account and debit a customer's account.
- Credit a general ledger account and debit another general ledger account.
- Debit a general ledger account and credit a customer's account.

This transaction can be used in several situations. For example, refunding a fee to a customer or receiving a payment from a person where there is no other obvious transaction to use. If the debit and credit currencies are different, the foreign exchange is calculated.

#### To debit or credit a general ledger account

- Navigate to the Teller Operations > General Ledger screen (Fast Tran 56).
- Select the Transaction Type. More information is available in the following table.

Field	Comment	
Correction - From Customer Account	A specific transaction type where you are transferring amount from a particular customer account. General Ledger account details are automatically populated and cannot be modified.	
Correction - To Customer Account	A specific transaction type where you are transferring amount to a particular customer account. General Ledger account details are automatically populated and cannot be modified.	
Customer Transfer to GL Account	A generic transaction type where you are transferring amount from a particular customer account. You need to type the General Ledger account details.	
GL Transfer to Customer Account	A generic transaction type where you are transferring amount to a particular customer account. You need to type the General Ledger account details.	
Inter GL Transfer	A generic transaction type where you are transferring amount between two General Ledgers. You need to type both the General Ledger account details.	
Refund Of Fee Or Charge	A specific transaction type where you are transferring amount to a particular customer account. General Ledger account details are automatically populated and cannot be modified.	

- Type the amount and complete the other fields.
- Click Submit.

### 8.8 Transferring Cash Between Cashboxes

You use the Cashbox Cash Transfer operation to transfer cash between two standard cashboxes. This allows you to manage the totals and denominations of cash within your cashbox.

Before you can make a cashbox transfer, both you and the other cashbox holder must log in and complete a Teller Start of Day operation. Both of you must have been assigned a cashbox in single mode, and allocated the cash in and cash out cash drawers. Both cashboxes must contain the currency that is required to be transferred. Both you and the other teller must authorize the transfer.

Siebel Branch Teller debits the transferring teller cashbox, and credits the receiving teller's cashbox with the cash and denomination amounts for the cash transfer. Siebel Branch Teller updates both your journal and the journal of the other teller with entries showing the amount of cash given or received.

#### To transfer cash between cashboxes

- Navigate to the Teller Operations > Cashbox Cash Transfer screen (Fast Tran 57).
- Select the Currency Box of the cashbox that the cash is to be transferred to or from.
- Click Transfer To Cashbox or Transfer From Cashbox depending on whether you want to transfer cash to or from the cashbox.

The Cashbox Cash Transfer screen is then displayed with the Transfer From and Transfer To fields containing the two cashbox IDs, and the Currency fields

- containing the currency of the selected currency box. The User ID field contains your user ID.
- **4.** Type the number of notes or coins for each denomination in the Number column, or the total in the Total column. If you enter the Number value, the corresponding Total field is automatically updated, or if you enter the Total value, the corresponding Number field is automatically updated.
  - The Total Notes, Total Coins, and Total Cash fields are then calculated by Siebel Branch Teller and displayed.
- 5. Type your password and make sure that the other teller types the user ID and password.
- Click Submit.

### 8.9 Transferring Checks Between Cashboxes

You use the Cashbox Check Transfer operation to transfer checks between two standard cashboxes, or between a standard cashbox and a vault cashbox.

Before you can make a cashbox check transfer, both you and the other cashbox holder must log in and complete a Teller Start of Day operation. Both cashboxes must contain the type of negotiable instrument that is required to be transferred. Both you and the other teller must authorize the transfer.

Siebel Branch Teller debits the transferring teller cashbox and credits the receiving teller's cashbox with the appropriate number of checks transferred. Siebel Branch Teller updates your journal and the journal of the other teller with appropriate entries.

#### To transfer a negotiable instrument between cashboxes

- Navigate to the Teller Operations > Cashbox Check Transfer screen (Fast Tran 58).
- Select a cashbox by selecting a negotiable instrument type.
- Click Transfer To or Transfer From depending on whether you want to transfer to or from the cashbox.
  - The Cashbox Check Transfer screen is then displayed with the Transfer From and Transfer To fields containing the two cashbox IDs. The User ID field contains your
- **4.** Type the start serial number and the end serial number for the checks.
- **5.** Click Add.
  - The negotiable instruments are added to the Negotiable Instrument Summary Table.
- **6.** Type your password and make sure that the other teller types the user ID and password.
- **7.** Click Submit.

### 8.10 Transferring Ownership of a Cashbox

You use the Cashbox Take Over operation to transfer the ownership of a shared cashbox to the non-primary holder of the cashbox. You must be the primary owner of the cashbox.

As part of the Cashbox Take Over operation, Siebel Branch Teller records the totals in each currency box in the cashbox and any difference between these totals and the system-generated totals.

When you transfer cashbox ownership, the non-primary owner becomes the primary owner of the cashbox, who is then responsible for cashbox cash totals and check batching. Siebel Branch Teller creates an entry in both of your journals.

#### To transfer ownership of a cashbox

- Navigate to the Teller Operations > Cashbox Take Over screen (Fast Tran 59).
- Do one of the following:
  - Type the number of notes or coins for each denomination in the Number column. If you enter the Number value, the corresponding Total field is automatically updated.
  - Type the total in the Total column. If you enter the Total value, the corresponding Number field is automatically updated.
- Click Confirm Totals to confirm if there is any difference between the system-generated totals and the actual totals.
  - Siebel Branch Teller calculates and updates the Total Notes, Total Coins, Total Cash, System Cash, and Difference fields.
- **4.** Select Next to enter the denomination details for the next currency box.
- 5. Repeat steps 2 through 4 for each currency box in the cashbox.
- Type your password, and make sure that the non-primary holder of the cashbox types the user ID and password.
- **7.** Click Submit.

### 8.11 Closing Open Batches of Checks

You use the Check Batch operation to close a batch of checks. As a result of check deposits, your bank (the collecting bank) must forward batches of checks to the bank where the funds are held (the paying bank) for reimbursement. Checks of a particular type and currency are held in batches in your cash drawer. When the number of checks in a batch reaches a certain limit, you must close the batch, and remove the checks from your cash drawer.

Siebel Branch Teller advises you when a batch is approaching the system limit, and again when the batch has reached the system limit. You must then close the batch, and in any case before a Teller End of Business Day or Teller End of Day operation.

#### To close an open batch of checks

- Navigate to the Teller Operations > Check Batch screen (Fast Tran 510).
- **2.** Select the check batch to close.
- Click Close Batch.
  - The Display Open Check Batches: Close Batch screen is displayed.
- Complete the Total Number of Items details. The fields are described in the following table.

Field	Comment
System	Displays the system total for the number of checks.
Teller	Type the number of checks in the batch you removed from your cash drawer.
Difference	Displays the difference, if any, between the System and Teller fields.

Complete the Total Item Amount details. The fields are described in the following table.

Field	Comment	
System	Displays the total amount for the batch.	
Teller	Type the total amount for the batch you removed from your cash drawer.	
Difference	Displays the difference, if any, between the System and Teller fields.	

- **6.** Click Batch Item Details, if required.
- 7. Check that the Difference fields both contain zero, and click Submit.

### 8.12 Transferring a Checkbook

You use the Checkbook Transfer operation to transfer a checkbook from your cashbox to the customer, or another branch vault.

Siebel Branch Teller updates the branch vault details, for example, date, and serial numbers.

#### To perform a checkbook transfer

- 1. Navigate to the Teller Operations > Checkbook Transfer screen (Fast Tran 511).
- Enter the account details if they are not already displayed; see Section 5.3.1, "Selecting the Account".

The checkbooks associated with the account are displayed in Check Transfer Details.

- **3.** Select one or more checkbooks in the list.
- **4.** Select the destination to which the checkbook is to be transferred:
  - Click Customer to transfer to the customer.
  - Click Other Branch and type a branch code to transfer to another branch vault.
- **5.** Enter your user ID and password.
- Click Submit.

### 8.13 Ending the Business Day

You use the Teller End of Business Day to carry on processing in the P.M. period, or to sign off for the day if you do not want to continue working during the P.M. period.

If you want to continue working during the P.M. period (click Go To Next Day), Siebel Branch Teller sets the processing date to the next business day, closes your current journal, and creates a journal for the new processing date.

If you do not carry on into the P.M. period, you must perform a Teller End of Day operation.

Before you can perform the Teller End of Business Day operation:

- The branch must be operating in AM/PM processing mode.
- The current processing date must be the same as the calendar date.
- There must be no open check batches.

#### To end the business day

- 1. Navigate to the Teller Operations > Teller End of Business Day screen (Fast Tran
- 2. If you want to continue working during the P.M. period, click Go To Next Day, and then click Submit.
  - If you are the primary holder of the cashbox, Siebel Branch Teller records the Teller End of Business Day currency totals in the cashbox. The operations home page is then displayed with the updated processing date. A user with the appropriate privileges, for example, a supervisor or branch manager, must then complete a Branch End of Business Day operation to allow transaction processing in the P.M. period. See, Section 8.13, "Ending the Business Day".
- 3. If you do not want to continue working during the P.M. period, click Final Sign Off for the Day, and then click Submit.

If you are the primary holder of the cashbox, Siebel Branch Teller records the Teller End of Business Day currency totals in the cashbox, and you then must perform the Teller End of Day operation, see Section 8.14, "Performing End-of-Day Processing".

### 8.14 Performing End-of-Day Processing

You use the Teller End operation to end your day.

If you are the primary owner of your cashbox, at the end of your day you must enter the denominations for each currency box in your cashbox to compare each of these totals with the system-generated totals. If there is a difference between these totals, you must rectify the difference before you can complete your end of day.

To resolve any difference you can view your journal for the current processing day and make the necessary reconciliation journal entries. If you cannot resolve the difference, a supervisor can authorize the balancing. This authorization resets the currency box denominations of the whole cashbox with the actual totals and completes the Teller End of Day process for you.

If you previously performed a Teller End of Business Day operation, Siebel Branch Teller has already created a journal for the current processing date and this is used for the next business day. Otherwise, Siebel Branch Teller closes your journal.

**Note:** If there are transactions that have been rejected after a period of offline processing, you must reconcile these transactions before you can complete the end-of-day processing. For more information, see Section 8.6, "Reconciling Failed Transactions".

#### To perform the end-of-day processing

1. Navigate to the Teller Operations > Teller End of Day screen (Fast Tran 513).

**Note:** If you are a cashless user and when you perform the Teller End of Day, you will receive the confirmation message asking whether you are sure to complete the teller end of day. If you click yes, you get another confirmation message whether you want to log out. If you click no, you will remain in Siebel Branch Teller. If you click yes, you will be logged out.

If you are the non-primary holder of your cashbox, the following message is displayed:

Are you sure you want to complete an end of day?

If you are the primary holder of your cashbox, the View Cash Details screen is displayed.

If you are the non-primary holder of your cashbox, click Yes.

You have now finished for the day.

- **3.** If you are the primary holder of your cashbox, do one of the following:
  - Type the number of notes or coins for each denomination in the Number column. If you enter the Number value, the corresponding Total field is automatically updated.
  - Type the total in the Total column. If you enter the Total value, the corresponding Number field is automatically updated.
- **4.** Click Confirm Totals.

The Total Cash, System Cash, and Difference fields are then calculated by Siebel Branch Teller and displayed.

**5.** If there is a difference in the totals, type an internal account number and branch code, and click Balance to post a balancing transaction. (You can click View Journal to help identify the cause of the difference.)

The Teller Settlement screen is displayed, depending on the nature of the difference in totals.

- Make a debit or credit against an internal account to offset the amount over or amount short. If the amount that needs to be balanced is beyond a certain threshold, your supervisor must authorize.
- **7.** Click Next to enter the details for the next currency.
- Repeat steps 3 through 7 for each currency in your cashbox.
- Click Submit.

### 8.15 Logging Out

You use the Logoff operation to log out of Siebel Branch Teller. If a Teller End Of Day operation is required, a message is displayed to prompt you to perform that operation. After you log out, Siebel Branch Teller closes.

#### To log out

- Navigate to the Teller Operations > Logoff screen (Fast Tran 514).
- Click Yes on the User Logoff screen.

# **Performing Branch Operations**

You use the Branch operations to start and end the branch day, set up cashboxes, and control the logging in of users.

The branch operations are available only to users with appropriate privileges.

You cannot perform the operations in a customer session.

This chapter covers the branch operations that you can perform. It includes the following topics:

- Section 9.1, "Starting the Branch Day"
- Section 9.2, "Searching Branch Information"
- Section 9.3, "Performing Cashbox Maintenance"
- Section 9.4, "Performing Vault Maintenance"
- Section 9.5, "Making External Cash Transfers"
- Section 9.6, "Making External Check Transfers"
- Section 9.7, "Resetting the Login Counter"
- Section 9.8, "Forcing Sign Off"
- Section 9.9, "Ending the Branch Business Day"
- Section 9.10, "Ending the Branch Day"

### 9.1 Starting the Branch Day

When your branch opens, you log in to Siebel Branch Teller, and confirm the branch processing date. You must have the required privileges to do so, for example, you must be a manager or a supervisor.

When you log in, Siebel Branch Teller displays the processing date and the cash totals from the previous day.

The way in which a Branch's current processing date is set depends on whether the branch is operating in AM processing mode or AM/PM processing mode:

- If the Branch is operating in AM processing mode, the current processing date is obtained from the system calendar.
- If the Branch is operating in AM/PM processing mode, the current processing date will have been set when the Branch undertook its previous Branch End of Business day operation.

After you confirm the processing date, creates a journal for you, and records a Branch Start of Day operation in the events log.

#### To start the branch day

- **1.** Log in to Oracle's Siebel Branch Teller application. Siebel Branch Teller displays the processing date and the cash totals from the previous closing position.
- If you are working in offline mode, check the processing date and enter the correct date if necessary.
- 3. Click Submit.

### 9.2 Searching Branch Information

You use the Branch Search operation to search for details about branches.

#### To search for branch information

- Navigate to the Branch Operations > Branch Search screen (Fast Tran 61).
- Select Branch Code, Branch Name, or Zip/Postal Code, and type the appropriate branch code, branch name, and zip or postal code.
- **3.** Click Search.

### 9.3 Performing Cashbox Maintenance

You use the Create Cashbox Maintenance to create a new cashbox and its associated currency boxes.

You can create a standard cashbox or a branch vault cashbox. You can create a cashbox in single or shared mode. You can create currency boxes for each of the currencies that your bank uses, and also negotiable instrument boxes for each negotiable instrument type.

To perform the operation, you must have the appropriate privileges to create a new cashbox.

Siebel Branch Teller creates the cashbox with the default currency box containing the base currency of your organization.

#### To create a cashbox

- Navigate to the Branch Operations > Cashbox Maintenance screen (Fast Tran 62).
- Click New. Complete the required fields.

Field	Comment
Cashbox Type	Select a cashbox type. The options include Branch Vault and Standard.
Cashbox Mode	Select Single or Shared.

- **3.** Click Currency Box.
- Select the check box for each currency in the cashbox, and click Submit. For each currency that you select, a currency box is created.
- If required, click Negotiable Instrument Box.

- **6.** Select a type from Negotiable Instrument Type, and then select the check box for each currency required. Repeat this step for all of your negotiable instruments. Finally, click Submit.
- 7. Click Submit to create the cashbox and its associated currency and negotiable instrument boxes.

You have an empty cashbox, therefore you must transfer cash and possibly negotiable instruments into the cashbox before you can perform financial transactions.

You can edit the cashbox details by going to Branch Operations > Cashbox Maintenance screen, selecting the cashbox, and clicking Amend. You can change the details and click Submit.

You can remove a cashbox by going to Branch Operations > Cashbox Maintenance screen, selecting a cashbox, and clicking Delete. You can confirm the deletion in the message box that appears.

### 9.4 Performing Vault Maintenance

You use the Vault Maintenance operation to transfer cash to or from the vault cashbox. Before you can make a transfer, both you and the other cashbox holder must log in and complete a Teller Start of Day operation. The other teller must be assigned a standard cashbox. Both cashboxes must contain the currency that is required to be transferred. Both you and the other teller must authorize the transfer.

Siebel Branch Teller debits or credits the respective cashboxes with the cash and denomination amounts for the cash transfer. Siebel Branch Teller updates your journal and the journal of the other teller with entries showing the amount of cash given or received.

#### To perform vault maintenance

- 1. Navigate to the Branch Operations > Vault Maintenance screen (Fast Tran 63).
- Select the Currency Box of the cashbox that the cash is to be transferred to or from.
- Click Transfer To or Transfer From depending on whether you want to transfer cash to or from the vault cashbox.

The Vault Denominations screen is then displayed with the Transfer From and Transfer To fields containing the two cashbox IDs, and the Currency fields containing the currency of the selected currency box. The User id field contains your user ID.

- **4.** Do one of the following:
  - Type the number of notes or coins for each denomination in the Number column. If you enter the Number value, the corresponding Total field is automatically updated.
  - Type the total in the Total column. If you enter the Total value, the corresponding Number field is automatically updated.

The Total Notes, Total Coins, and Total Cash fields are then calculated by Siebel Branch Teller and displayed.

- Type your password and make sure that the other teller types the user ID and password.
- 6. Click Submit.

### 9.5 Making External Cash Transfers

You use the External Cash Transfer operation to transfer inward and outward cash to and from your branch. Inward cash is cash supplied to the branch from an authorized external party. Outward cash is funds removed to an authorized external party. You can make such transfers only if you are an authorized branch user; that is, a vault administrator, supervisor, or manager.

To perform the operation, you must have completed a Teller Start of Day operation, and you must own a vault cashbox.

Siebel Branch Teller updates your journal and vault cashbox, and prints a transactions receipt.

#### To make an external cash transfer

- 1. Navigate to the Branch Operations > External Cash Transfer screen (Fast Tran 64). A list of your assigned cashboxes is displayed, with a line for each of the currency boxes in each of the cashboxes.
- **2.** Select a currency box.
- Click Transfer To or Transfer From, as applicable.

The External Cash Transfer To Vault screen or the External Cash Transfer From Vault screen is then displayed.

- **4.** Do one of the following:
  - Type the number of notes or coins for each denomination in the Number column. If you enter the Number value, the corresponding Total field is automatically updated.
  - Type the total in the Total column. If you enter the Total value, the corresponding Number field is automatically updated.
    - Siebel Branch Teller calculates and displays the values for Total Notes, Total Coins, and Total Cash.
- **5.** Enter your supervisor user ID and password in the respective Authorization Details fields.
- **6.** Click Submit.

### 9.6 Making External Check Transfers

You use the External Negotiable Instrument Transfer operation to transfer negotiable instruments to or from two different branch vault cashboxes, and to and from an external supplier.

You can make such transfers only if you are an authorized branch user; that is, a vault administrator, supervisor, or manager. You must have completed a Teller Start of Day operation, and you must own a vault cashbox. Siebel Branch Teller updates your journal and vault cashbox.

#### To make an external negotiable instrument transfer

- Navigate to the Branch Operations > External Check Transfer screen (Fast Tran 65). A list of your assigned cashboxes is displayed, with a line for each of the currency boxes in each of the cashboxes.
- **2.** Select a negotiable instrument currency box.

**3.** Click Transfer To or Transfer From, as applicable.

The appropriate screen is then displayed. There are tabs for Traveler's Checks and for Drafts/Cashiers Checks/Money Orders. If you had clicked Transfer From, then there is a tab for Customer Checks as well.

Field	Comment
Draft, Money Order, Cashiers Check	Select one of these radio buttons, depending on the negotiable instrument type.
Traveler's Check	Type the traveler check details if you had clicked Transfer To. Type the traveler check details and select the source details if you had clicked Transfer From.
Customer's Check	Type the check details and select the source details.
Denomination	Select the denomination for the travelers check.
Currency Details	Displays the currency for the negotiable instrument.
Start Serial Number	Type the start serial number for the negotiable instrument.
End Serial Number	Type the end serial number for the negotiable instruments.

#### **4.** Click Add.

The negotiable instruments are added to the Negotiable Instrument Summary Table.

- Enter your supervisor user ID and password in the respective Authorization Details fields.
- Click Submit.

## 9.7 Resetting the Login Counter

You use the Failed Logon Counter operation to reset the login counter for a teller who has exceeded the limit for login attempts, and is therefore locked out of Siebel Branch Teller.

If a user unsuccessfully tries to log in more than a certain number of times, that user is locked out of Siebel Branch Teller. The operation resets the login counter and allows the user to log in.

To perform the operation, you must have appropriate supervisor privileges.

#### To reset the login counter

- Navigate to the Branch Operations > Failed Logon Counter (Fast Tran 66).
- Select the user ID for the user whose login counter you want to reset.
- Enter your supervisor user ID and password in the respective Authorization Details fields.
- **4.** Click Reset Counter. The user can now log in to Siebel Branch Teller.

### 9.8 Forcing Sign Off

You use the Forced Sign Off operation to allow a teller to log in when Siebel Branch Teller considers them to be already logged in. To perform the operation, you must be a supervisor.

When you force sign off for a teller, Siebel Branch Teller considers the teller to be temporarily logged out. The teller is not required to perform a Teller End of Day operation at this time, but the cashbox remains assigned to the teller.

#### To force sign off

- 1. Navigate to the Branch Operations > Forced Sign Off screen (Fast Tran 67). The list of users who are currently logged in is displayed (your details are not displayed).
- Select the user for which you want to force sign off.
- 3. Enter your supervisor user ID and password in the respective Authorization Details fields.
- **4.** Click Forced Signoff.

### 9.9 Ending the Branch Business Day

You use the Branch End of Business Day operation when the branch is operating in AM/PM processing mode, and you want to continue processing during the P.M. period. This allows you to set the branch processing date to the next business day, which facilitates timely clearing house settlement.

Before you perform the operation, any teller who wants to continue operating during the P.M. period must have performed a Teller End of Business Day - Go To Next Day operation. Any teller not continuing during the P.M. period must have performed a Teller End of Business Day - Final Sign Off operation.

Siebel Branch Teller records a Branch End of Business Day operation in the events log.

#### To end the branch business day

- Navigate to the Branch Operations > Branch End of Business Day screen (Fast Tran 68).
- **2.** Click Submit.

The operations home page is displayed with the processing date set to the next business day.

### 9.10 Ending the Branch Day

You use the Branch End of Day operation when you close the branch at the end of the day. The processing date and the cash totals for the closing position are displayed. You must have the required privileges to perform the operations, for example, you must be a manager or a supervisor.

Before you can perform the operation, all tellers must have completed their day, and you must be the last person to complete your day. If the branch is operating in AM/PM processing mode, someone must have performed a Branch End of Business Day operation.

After you complete the operation, Siebel Branch Teller records a Branch End of Day operation in the events log. If you have completed transaction processing for the current processing date, your journal is closed for the current processing date.

**Note:** Do not confuse this operation with the Branch End of Business Day operation. See, Section 8.13, "Ending the Business Day".

#### To end the branch day

- Navigate to the Branch Operations > Branch End of Day screen (Fast Tran 69). Siebel Branch Teller displays the processing date and the cash totals for the closing position.
- Click Submit.

After you complete the operation, you can then log out.